CANADIAN DEVELOPMENT ASSISTANCE

A SELECTED BIBLIOGRAPHY 1950~77

SHIRLEY B. SEWARD & HELEN JANSSEN



Published by the International Development Research Centre in collaboration with the Norman Paterson School of International Affairs, Carleton University

IDRC-113e

The International Development Research Centre is a public corporation created by the Parliament of Canada in 1970 to support research designed to adapt science and technology to the needs of developing countries. The Centre's activity is concentrated in five sectors: agriculture, food and nutrition sciences; health sciences; information sciences; social sciences; and communications. IDRC is financed solely by the Government of Canada; its policies, however, are set by an international Board of Governors. The Centre's headquarters are in Ottawa, Canada. Regional offices are located in Africa, Asia, Latin America, and the Middle East.

©1978 International Development Research Centre Postal Address: Box 8500, Ottawa, Canada K1G 3H9

Head Office: 60 Queen Street, Ottawa

Seward, S. B.
Janssen, H.
IDRC, Ottawa
Carleton University, Ottawa. Norman Paterson School of International
Affairs

IDRC-113e Canadian development assistance: a selected bibliography 1950-77. Ottawa, Ont., IDRC, 1978. 62p.

/IDRC publication / . / Bibliography / on / Canada / 's / foreign aid / / aid programme / s and policies from 1950 to 1977.

UDC: 016:341.232(71) ISBN: 0-88936-187-8

Microfiche edition available

Canadian Development Assistance: A Selected Bibliography 1950-77*

Shirley B. Seward and Helen Janssen

Edited by: Jane Beaumont

Published by the International Development Research Centre in collaboration with the Norman Paterson School of International Affairs Carleton University

^{*}This article has also been designated as No. 6 in the Norman Paterson School of International Affairs Bibliography Series, Carleton University, Ottawa, Canada.

Contents

Preface	3
Acknowledgments	4
Introduction	5
Chapter 1. General Works on Canadian Development Assistance	9
Chapter 2. Regional Aid Programs	33
2.1 Asia	33
2.2 Caribbean	39
2.3 Africa — General	ю
2.3.1 Commonwealth Africa	1
2.3.2 Francophone Africa	12
2.4 Latin America	13
Chapter 3. Official Publications	15
3.1 Canadian Government	15
3.2 International Organizations	52
Chapter 4. Bibliographies and Reference Works	57
Author and Institutional Index	ţ

Preface

This Bibliography was initiated by Shirley B. Seward, in collaboration with the late A. F. W. Plumptre, as part of a broader research endeavour examining Canada's relations with the developing world. Recognizing the importance of the activity, the Norman Paterson School of International Affairs at Carleton University provided editorial guidance and research assistance in order to complete the work.

The International Development Research Centre, desiring both to participate in the undertaking and to ensure wider distribution of the Bibliography, suggested a joint Norman Paterson School of International Affairs/IDRC publication, and we are happy to have collaborated in this work. The senior author is now a program officer with the Social Sciences Division of IDRC and the junior author is a graduate student at the Norman Paterson School of International Affairs at Carleton University. The text was edited by Jane Beaumont, the Coordinator of the Resource Centre at the Norman Paterson School of International Affairs at Carleton University.

Reginald MacIntyre

Director, Communications Division International Development Research Centre

Acknowledgments

Many individuals have assisted in the compilation and verification of the Bibliography. In particular, the authors would like to thank Donald Page, who permitted free access to his references on development aid; Boyd McBride, who provided research assistance during the early stages; the reference librarians at the Department of External Affairs, CIDA, IDRC, Carleton University, and the National Library; and Bruce Rawlins of Banfield who designed the cover.

Introduction

One of the most striking phenomena of the third quarter of the Twentieth Century has been the conscious and deliberate attempt to stimulate economic and social development internationally through the transfer of resources from affluent to developing nations.

The historical precedents influencing the initiation of international development assistance in the fifties were twofold: the termination of European colonialism and the concomitant emergence of newly independent nations; and the optimism springing from the incredible success of Marshall Plan capital assistance in restoring the war-torn economies of Europe. Just as the concept of transfer of resources had been validated in the case of European countries, it was hoped that a similar initiative would stimulate economic development in the Third World.

The history of Canada's involvement in international development assistance since 1950 is evident in the literature referred to in this Bibliography. Chapter One contains a wide variety of general works on Canadian aid programs, policies, and issues. The literature is concerned with a range of topics including the strategies developed by the Canadian International Development Agency, the administration of aid, trade versus aid, bilateral versus multilateral assistance, tied versus untied aid, and many other issues. Chapter Two contains five sections dealing with regional aid programs in Asia. the Caribbean, Commonwealth Africa, Francophone Africa, and Latin America. This section contains articles about bilateral and multilateral aid programs and projects in each region of the world. Chapter Three contains information about Government and official publications. The first section includes the main Canadian Government agencies and departments that have an input into foreign aid policies and programs. Listed in this section are the publications that regularly contain articles about development aid. Particularly interesting are the Parliamentary Debates and the Statements and Speeches published by External Affairs. These two publications are rich sources of information on perceptions of various actors with respect to Canadian foreign policy. The second section is devoted to multilateral institutions through which Canada channels aid funds. This section does not contain articles about multilateral aid programs as these are included in the General and Regional sections of the Bibliography; rather, a listing is made of those documents published by the multilateral institutions that include reference to Canada's contributions. Finally, Chapter Four lists the main bibliographies and reference works that have direct relevance to Canadian foreign policy and development assistance.

Within the international forum, Canada has played a modest but important role in development assistance. From the outset, Canada has disbursed both bilateral and multilateral aid. The evolution of Canada's aid programs reflects both Canadian motivations underlying development assistance and Canada's special relations with various regions of the world.

Canada's first bilateral program was established in Asia with the inception of the Colombo Plan for Co-operation and Economic Development in South and Southeast Asia. Conceived at a meeting of the Commonwealth Foreign Ministers in Colombo in January 1950, the Colombo Plan became an umbrella for bilateral aid from several donors, including Canada, to newly independent Commonwealth countries of Asia. The Plan later spread to include the participation of non-Commonwealth donors and recipients. Assistance to Asia today constitutes Canada's largest bilateral aid program.

In 1958, Canadian assistance expanded into the Commonwealth Caribbean. Once again, a program was established to assist newly independent members of the Commonwealth as they united in the fledgling West Indies Federation. After the Federation collapsed in 1962, the Canadian aid program continued and expanded as separate programs to the individual islands.

The third region of the world in which Canada became involved in development aid was Commonwealth Africa. Beginning in the fifties, a number of Anglophone African countries gained independence from Britain and became members of the Commonwealth. At the Commonwealth Prime Ministers' Meeting in 1960, the Special Commonwealth African Assistance Plan was conceived. Following the pattern established in Asia and the Caribbean, Canada joined the scheme to assist newly emerging nations.

The wave of independence was also very much in evidence in Francophone Africa. In order to provide assistance to the former colonies of France and Belgium, the Secretary of State for External Affairs announced the establishment of a Francophone African Educational Program in 1961. The Francophone program expanded considerably after the Chevirer Mission to Africa in 1968 and the Foreign Policy Review in 1970, and has been receiving considerable attention ever since.

The last area of the world in which Canada established a bilateral aid program was Latin America. Through a rather unusual arrangement, Canada utilized the Inter-American Development Bank to channel bilateral funds to Latin American countries. Although Canada was not a member of the Bank until 1972, a special Canadian fund was started in 1964 to support projects that were suggested by recipient countries, assessed by the Bank, and approved by CIDA. Canada further confirmed its commitment to the region by establishing a bilateral technical assistance program in 1971, and by becoming a member of the Inter-American Development Bank in 1972. Both of these moves were products of the 1970 Foreign Policy Review.

In addition to its bilateral aid programs, Canada channels funds through a variety of multilateral organizations including the United Nations family, the World Bank Group, the Regional Development Banks, and many other smaller programs. In recent years Canadian funds flowing through multilateral channels have ranged between 15 and 20 percent of the total aid program. In its 1970 Foreign Policy Review, the Canadian Government expressed its intention to increase this proportion to about 25 percent.

The United Nations family can be divided into two major groups: the specialized agencies and the social and economic programs. The specialized agencies include the FAO, ILO, UNESCO, WHO, and WIPO. The UN social and economic programs include the UNDP, UNHCR, UNICEF, UNFPA, UNRWA, UNITAR, and the joint UN/FAO World Food

Program. 1 The UNDP is regarded as the largest and most important preinvestment and technical assistance agency in the world, and receives the bulk of Canadian multilateral funds within the UN family.

The largest recipient of Canadian multilateral funds is the International Development Association (IDA), the arm of the World Bank that lends funds on concessional terms to the least developed countries.

Canada is a member of the Asian, Caribbean, and Inter-American Development Banks. Furthermore, although membership in the African Development Bank is restricted to African countries, Canada helped establish and contributes to the African Development Fund. In addition to the four regional development banks, Canada also contributes to a subregional financial institution, the Andean Development Corporation.

Finally, the multilateral program is extended to a number of other organizations and institutions, including the International Agricultural Research Centres, the Commonwealth Fund for Technical Cooperation (CFTC) and the Fond international de coopération universitaire (FICU), to name but a few.

The bilateral and multilateral aid programs are developed and administered by Canada's official aid agency, the Canadian International Development Agency (CIDA). Established in 1968, CIDA succeeded the earlier External Aid Office. Since that time, the Agency and the Canadian Government have undertaken a series of reviews and evaluations of Canadian aid programs, policies, and strategies.

One of the most significant and innovative outcomes of this soul-searching was the establishment of the International Development Research Centre (IDRC). The Centre was created as a public corporation by the Parliament of Canada in 1970, to support research to adapt science and technology to the needs of developing countries. In fulfilling its mandate, IDRC has concentrated its efforts on supporting research designed and carried out by indigenous researchers within the developing country context. Although IDRC is financed solely by the Government of Canada, its policies are set by an international Board of Governors.

Canadian motivations underlying participation in international development assistance programs consist of a complex mix of political, economic, and humanitarian factors, and have varied over time and from region to region.

From a political perspective, Canada has had much to gain in terms of its international reputation, particularly within the Commonwealth, by assisting newly independent Commonwealth nations in Asia, the Caribbean, and Africa. In the early days of the Colombo Plan in Asia, this motive was reinforced by the Communist-West struggle, which was beginning to assume importance at that time. The donor community, including Canada, viewed it as particularly critical to assist in raising the standard of living in Asian countries, thereby making the Communist alternative less attractive.

The political motivation has also been fundamentally important within the domestic context. It has been argued that the aid programs in Francophone

¹Many classifications of the UN system are utilized. This particular type of breakdown is usefully employed in *Canada and the United Nations*, 1945-1975, Department of External Affairs, 1977.

Africa, and to a lesser extent in Latin America, were initiated in part to demonstrate to French Canada that the federal government also had an interest in non-Commonwealth, non-Anglophone countries.

The economic motivation in developing aid programs has been related to Canada's incentive to open up foreign markets for Canadian goods and services. This motive was probably most evident in the Caribbean, where trade relations with Canada had existed since the seventeenth century, and Latin America, where the potential markets were perceived as almost limitless.

Finally, underlying the political and economic motives is the basic humanitarian concern for alleviating poverty in all parts of the world, thereby contributing to the goals of international justice and stability.

Shirley B. Seward
October 1978

Chapter 1

General Works on Canadian Development Assistance

A

- "Aid, Trade and Development." International Journal, 25 (Winter 1969-70).
- "Aide à l'étranger: Augmentation constante." Sept-Jours, 2 (14-20 janvier 1968), p.15.
- "Aide extérieure: Le Canada se lance." Sept-Jours, 2 (24 août 1968), pp. 13-14.
- "L'aide en 1972." Coopération Canada, 11 (novembre-décembre 1973), pp. 12-13 (Eng. "Aid Flows During 1972").
- Anglin, Douglas G. "Canada's External Assistance Programme." *International Journal*, 9 (Summer 1954), pp. 193-207.
- Anglin, Douglas G. "The Canadian Post-War Contribution to International Assistance." *International Journal*, 9 (Autumn 1954), pp. 264-74.
- "Apportionment of Foreign Aid Debated." External Affairs, 23 (August 1971), pp. 297-303.
- Armstrong, Gregory. "Aid Policies as a Reflection of Canadian Domestic Concerns: From Colombo to CIDA." *International Perspectives*, (March-April 1975), pp. 44-8.
- Armstrong, Gregory. "Les politiques d'aide: reflet des préoccupations fédérales." Perspectives internationales, (mars-avril 1975), pp. 49-53.
- "Au secours!" Coopération Canada, 13 (mars-avril 1974), pp. 16-22 (Eng. "Emergency!").
- "Avec l'aide privée bénévole, redonner à la coopération le sens d'une grande aventure humaine." Coopération Canada, 4 (septembre-octobre), pp. 10-15.

В

- Banff Conference on World Affairs, 7th, 1969. Imperatives for Development.

 Toronto: Canadian Institute of International Affairs, 1969.
- Barber, Clarence L. "More Aid to Underdeveloped Areas?" *International Journal*, 10 (Summer 1955), pp. 198-209.
- Barkway, Michael. "Canada's Weakest Spot." Saturday Night, 567 (May 24, 1952), pp. 2-3.
- Barrett, F. Michael; and Malcolm Fitz-Earle. "Should Canada Withhold Aid from Developing Countries Lacking Population Control?" Science Forum, 6 (April 1973), pp. 3-6.

- Baudouin, Marc. "Un nouvel ordre international le défi lancé par le Tiers-Monde." Coopération Canada, (novembre-décembre 1974), pp. 3-10 (Eng. "Towards the Year 2000: The Challenge from the Poor Nations").
- Bauer, P. T. "Commodity Agreements: Aid or Trade?" *International Journal*, 29 (Autumn 1974), pp. 610-18.
- Beamish, Royd E. "Foreign Aid: Why, What, How Much?" Canadian Business, 41 (October 1968), pp. 70-76; 41 (November 1968), pp. 64-72.
- Becker, Jane. "Report from the Frontiers of Friendship." *Maclean's*, 77 (June 6, 1964), pp. 18 ff.
- Belshaw, Cyril S. "Training and Recruitment: Some Principles of International Aid." *International Journal*, 18 (Winter 1962-63), pp. 43-57.
- Best, Jack. "Recherches avec le Tiers-Monde." Coopération Canada, 1 (mars-avril 1972), pp. 18-21 (Eng. "Lean and Professional Group Tackles Aid Research").
- Bled, Cynthia E. "Bilateral Foreign Aid: Canada." Broadcasting in Africa: A Continental Survey of Radio and Television. Edited by Sydney W. Head. Philadelphia, Pa: Temple University Press, 1974, pp. 240-7.
- Brault, Marcel. "Apport de la province de Québec aux pays en voie de développement." Canadian Educational and Research Digest, 7 (June 1967), pp. 147-54.
- Brecher, Irving. "Canada's Foreign Economic Aid." *Canadian Banker*, 69 (Winter 1962), pp. 39-50.
- Brecher, Irving. Canada's International Development Research Centre. Montreal: McGill University, Centre for Developing Area-studies, 1970 (Reprint Series, No. 18).
- Brecher, Irving. "The Continuing Challenge of International Development: A Canadian Perspective." Queen's Quarterly, 82 (Autumn 1975), pp. 323-43.
- Brewin, F. Andrew. "Canadian Economic Assistance to Under-Developed Areas." *International Journal*, 5 (Autumn 1950), pp. 304-14.
- Briant, Peter C. Canada's External Aid Program. Montreal: Private Planning Association of Canada, 1965.
- British Columbia, Ministry of Agriculture. Agricultural Aid to Developing Countries and World Disaster Areas Fund: 1972 Report to the Legislature of the Province of British Columbia. Victoria, B.C.: Department of Agriculture, 1972.
- British Columbia, Ministry of Agriculture. Agricultural Aid to Developing Countries and World Areas Disaster Fund: 1974 Report to the Legislature of the Province of British Columbia. Victoria, B.C.: Department of Agriculture, 1974.
- "Building the Future." Saturday Night, 68 (October 11, 1952), pp. 5-6.
- Burn, Henry Pelham. "Development Aid: End of an Era?" The Canadian Banker and ICB Review, 83 (September-October 1976), pp. 32-4.

- Bursey, M. B. "Greece: Development Projects Often Need Foreign Consultants." Foreign Trade, 130 (December 7, 1968), pp. 26-7.
- Burton, Ian. "Foreign Aid Telling It Like It Is." Canadian Forum, 48 (June 1968), pp. 50-1.
- Byers, R. B. "International Development Assistance." Canadian Annual Review of Politics and Public Affairs 1971. Edited by John Saywell. Toronto: University of Toronto Press, 1972, pp. 276-84.
- Byers, R. B. "International Development Assistance." Canadian Annual Review of Politics and Public Affairs 1972. Edited by John Saywell. Toronto: University of Toronto Press, 1974, pp. 290-4.
- Byers, R. B. "International Development Assistance." Canadian Annual Review of Politics and Public Affairs 1973. Edited by John Saywell. Toronto: University of Toronto Press, 1974, pp. 257-61.
- Byers, R. B. "International Development Assistance." Canadian Annual Review of Politics and Public Affairs 1974. Edited by John Saywell. Toronto: University of Toronto Press, 1975, pp. 322-6.
- Byers, R. B. "International Development Assistance." Canadian Annual Review of Politics and Public Affairs 1975. Edited by John Saywell. Toronto: University of Toronto Press, 1976, pp. 277-82.
- Byers, R. B. "The Third World." Canadian Annual Review of Politics and Public Affairs 1971. Edited by John Saywell. Toronto: University of Toronto Press, 1972, pp. 265-7.
- Byers, R. B. "The Third World." Canadian Annual Review of Politics and Public Affairs 1972. Edited by John Saywell. Toronto: University of Toronto Press, 1974, pp. 294-304.
- Byers, R. B. "The Third World." *Canadian Annual Review of Politics and Public Affairs 1973*. Edited by John Saywell. Toronto: University of Toronto Press, 1974, pp. 261-72.
- Byers, R. B. "The Third World." Canadian Annual Review of Politics and Public Affairs 1974. Edited by John Saywell. Toronto: University of Toronto Press, 1975, pp. 326-32.
- Byers, R. B. "The Third World." Canadian Annual Review of Politics and Public Affairs 1975. Edited by John Saywell. Toronto: University of Toronto Press, 1976, pp. 265-77.
- Byers, T. J. Foreign Resources and Economic Development. A Symposium on the Report of the Pearson Commission. London: Frank Cass, 1972.

 \mathbf{C}

- "Canada Aids Flood-Devastated Italy." External Affairs, 19 (May 1967), p. 181.
- "Canada Assists Turkish Communications." External Affairs, 19 (September 1967), pp. 365-6.
- "Canada Co-operating in Technical Assistance Programmes." External Affairs, 5 (November 1953), pp. 327-8.

- "Canada Gives \$100,000 to Refugee Fund." External Affairs, 5 (April 1953), p. 130.
- "Canada Helps Feed the Hungry." External Affairs, 18 (May 1966), pp. 210-11.
- "Canada Should Give More Foreign Aid." Canadian Commentator, 7 (February 1963), p. 22.
- "Canada and Technical Assistance to Under-Developed Countries." External Affairs, 4 (March 1952), pp. 90-5.
- "Canada and UNICEF: Twenty-five Years of Action." External Affairs, 23 (November 1971), pp. 451-2.
- "Canada and the Specialized Agencies." External Affairs, 7 (April 1955), pp. 116-21.
- "Canada and the World Health Organization." External Affairs, 20 (September 1968), pp. 389-92.
- Canada, Department of External Affairs. Canada and the United Nations 1945-1975.

 Ottawa: Ministry of Supply and Services, 1977, pp. 107-45.
- Canada, Department of External Affairs. Canada's Aid to Developing Countries.
 Ottawa: October 1976 (Reference Papers, No. 86).
- Canada, Department of External Affairs. Canada's Financial Contribution to the United Nations Family of Organizations. Ottawa: November 1974 (Reference Papers No. 93).
- Canada, Department of External Affairs. Foreign Policy for Canadians. Ottawa:
 Queen's Printer, 1970.
 Six Booklets: Foreign Policy for Canadians; Europe; International Development; Latin America; Pacific; United Nations.
- Canada, Department of External Affairs. The U.N. Food and Agriculture Organization. Ottawa: March 1974 (Reference Papers No. 121), pp. 7-8.
- Canada, Department of Industry, Trade and Commerce, International Financing Branch. *International Development and the Canadian Economy*. Ottawa: December 1968.
- Canada, External Aid Office. Canada: Non-Governmental Agencies in International Aid and Development: 1966-67. Ottawa: 1967.
- "Canada's Agricultural Aid Programmes." External Affairs, 17 (December 1965), pp. 538-9.
- "Canada's 1965 Contribution to Refugee Finances." External Affairs, 17 (October 1965), pp. 458-9.
- "Canada's Contributions to the United Nations." External Affairs, 9 (January 1957), pp. 31-4.
- "Canada's Contribution to the United Nations." External Affairs, 14 (December 1962), pp. 361-74.
- "Canada's Educational Aid Programme." External Affairs, 18 (January 1966), pp. 31-2.
- "Canada's External Aid Office: Annual Report, 1964-65." External Affairs, 17

- (August 1965), pp. 368-9.
- "Canada's Nuclear Energy Program." External Affairs, 22 (November 1970), pp. 389-92.
- "Canada's Post-War Financial Assistance Abroad." External Affairs, 8 (September 1956), pp. 254-6.
- Canadian Church Observers. "The World Food Conference: Canada's Contribution." Canadian Forum, 54 (January 1975), pp. 9-12.
- Canadian Council for International Cooperation. Canadian Youth Organizations and International Development. Ottawa: July 1970.
- Canadian Council for International Cooperation. Community Encounter:

 Programme Models for the Community. Ottawa: 1971.
- Canadian Council for International Cooperation. The Pearson Report: Dossier. Ottawa: 1970.
- Canadian Council for International Cooperation. Training for Service: Orientation and Training Courses for Work in Developing Countries. Compiled and edited by Edward W. Devlin. Ottawa: 1969.
- Canadian Council for International Cooperation. Youth on Development: Canadian Youth on International Aid and Development. Ottawa: 1971.
- "Canadian Executive Service Overseas." The Canadian Personnel and Industrial Relations Journal, 20 (January 1973), pp. 10-1.
- "Canadian External Aid." Commonwealth Survey, 11 (March 1965), p. 312.
- "Canadian Flour for Greece." External Affairs, 4 (August 1952), pp. 270-4.
- "Canadian Food Gift to Greece." External Affairs, 15 (October 1963), p. 367.
- "Canadian Foreign Aid: A Perspective on Policy and a Look at the Challenges Ahead." Bank of Nova Scotia Monthly Review, May 1977.
- "Canadian Foreign Aid and the C.I.A." *Canadian Dimension*, 5 (April-May 1968), pp. 28-31.
- Canadian Institute of International Affairs. International Development and Assistance: An Aid to Study Groups. Toronto: 1970.
- Canadian International Development Agency. Background Paper for Development Assistance Policy: Review Conference May 16, 17 & 18, 1969. Ottawa: 1969. (CIDA Policy Review)
- Canadian International Development Agency. Canada and the Developing World 1970-71. Ottawa: n.d.
- Canadian International Development Agency. Canada: Strategy for International Development Cooperation 1975-1980. Ottawa: September 1975.
- Canadian International Development Agency. Canadian Goods and Services for the Developing World. Ottawa: n.d.
- Canadian International Development Agency. The Canadian International Development Agency. Ottawa: January 1971.

- Canadian International Development Agency. CIDA Commitments and Disbursements. Ottawa: February 1971.
- Canadian International Development Agency. CIDA Handbook. Ottawa: n.d.
- Canadian International Development Agency. Examples of Problems in Development Cooperation. Ottawa: April 10, 1975.
- Canadian International Development Agency. Infrastructure and Environment: Telecommunications, Energy, Transportation, Mining, Environment. Ottawa: 1976. (Sectoral Guidelines 3)
- Canadian International Development Agency. Rural Development and Renewable Resources: Rural Development, Agriculture, Rural Water Development, Forestry, Fisheries. Ottawa: 1976. (Sectoral Guidelines 1)
- Canadian International Development Agency. Social Development and Community Services: Education, Health, Population, Housing, Social Communications, Cooperatives. Ottawa: 1976. (Sectoral Guidelines 2)
- Canadian International Development Agency. Taking Stock: A Review of CIDA Activities 1970-1974. Ottawa: 1974.
- Canadian International Development Agency, Policy Branch. *Aid Untying*. Ottawa: November 13, 1973.
- "The Canadian International Development Agency." External Affairs, 20 (November 1968), pp. 469-74.
- "Canadian International Development Agency: Annual Report 1967-68." *Industrial Canada*, 69 (December 1968), pp. 63-4.
- "Canadian Labour Congress Work in Developing Countries." Labour Gazette, 72, (May 1972), pp. 223-4.
- Canadian News Synthesis Project. Canadian Press Coverage of the Food Debate: An Analysis. Toronto: February, 1975.
- "Canadian Participation in Technical Assistance Programmes." External Affairs, 3 (January 1951), pp. 18-20.
- "Canadian Teachers Serving Abroad." External Affairs, 19 (October 1967), pp. 434-5
- "Canadian Technical and Educational Assistance: Teacher, University and Adviser Programmes." External Affairs, 17 (July 1965), pp. 292-9.
- "Canadian Technical and Educational Assistance Training Programmes." External Affairs, 17 (April 1965), pp. 131-8.
- Canadian University Service Overseas. Readings in Development, 3 vols. Compiled and edited by Robert D. H. Sallery, Mary L. McDonald, Paul G. Duchesne. Ottawa: 1969.
- "Canadians Abroad: The Paunch Corps." Time, 92 (September 1968), p. 15.
- Cavell, R. G. Nik. "A Bold New Design to Halt the Waste of Billions in Foreign Aid." *Maclean's*, 76 (April 6, 1963), pp. 20 ff.
- Cermakian, Jean. "La géographie de l'aide extérieure du Canada." The Canadian

- Geographer.-Le Géographe Canadien, 13 (Summer été 1969), pp. 163-8.
- Charles, K. J. "Foreign Aid What's In It for US?" Canadian Dimension, 3 (March April 1966), pp. 43-5.
- "CIDA to Aid Overseas Investment." Foreign Trade, 134 (December 5, 1970), p. 9.
- "CIDA Lines of Credit (as of November, 1973)." Canada Commerce, 138 (February 1974), p. 26.
- "CIDC: Canada's Move to Become the First 'International' Nation." Executive, 11 (July-August 1969), pp. 20-23.
- Clarkson, Stephen. "Canada's Role in Long-Term World Perspective: A Proposal for a New Initiative." Behind the Headlines, 28 (May 1969), pp. 1-10.
- Clarkson, Stephen (ed.). An Independent Foreign Policy for Canada? Toronto: McClelland & Stewart, 1968, pp. 226-52.
- "CLC Calls for One Per Cent Minimum for External Aid." Canadian Labour, 15 (April 1970), p. 20.
- Cohn, Theodore. "Food Surpluses and Canadian Food Aid." Canadian Public Policy, 3 (Spring 1977), pp. 141-54.
- Colaço, Francis X. Economic and Political Considerations and the Flow of Official Resources to Developing Countries. Paris: Development Centre of the Organisation for Economic Co-operation and Development 1973, pp. 52-7. (Development Centre Studies)
- Commonwealth Secretariat. Aid and the Commonwealth 1973: Report by the Commonwealth Secretariat. London: 1975, pp. 17-21.
- Couchiching Conference of the Canadian Institute on Public Affairs in Conjunction with the Canadian Broadcasting Corporation, 25th, 1956. Texts of Addresses delivered at the 25th Annual Couchiching Conference, August 4th-11th, 1956. Toronto: Canadian Institute on Public Affairs, 1956.
- Couchiching Conference of the Canadian Institute on Public Affairs, 44th, 1975. Canada and the Third World: What are the Choices? Toronto: Yorkminster Publishing Ltd., 1975.
- Cox, Terrance. "New Roles for Canada's Overseas Volunteers." Saturday Night, 89 (June 1974), pp. 24-8.
- Cunningham, George. "Australia and Canada as Aid Givers: Putting Over the Pearson Report." Round Table, 60 (April 1970), pp. 165-71.
- Cunningham, George. The Management of Aid Agencies: Donor Structures and Procedures for the Administration of Aid to Developing Countries. London: Croom Helm Ltd. in association with The Overseas Development Institute, 1974, pp. 119-33.

D

- "Dans le Tiers-Monde la médecine préventive tire le meilleur parti de ressources limitées." Coopération Canada, 3 (juillet-août 1972), pp. 27-30.
- de Bondt, John. "Canada's Aid to Developing Countries." Canadian Geographical Journal, 85 (November 1972), pp. 164-71.

- de Bondt, John. "Doing Business Through CIDA." Canada Commerce, 136 (July 1972), pp. 7-10.
- "Development Assistance Committee." External Affairs, 14 (October 1962), pp. 290-2.
- "Development Assistance Group." External Affairs, 13 (June 1961), pp. 217-19.
- "Development Assistance Group." External Affairs, 13 (September 1961), pp. 298-9.
- Devlin, Edward W. "Voluntary Agencies and World Development." External Affairs, 20 (August 1968), pp. 325-34.
- "Distribution of Canadian School Supplies in Greece." External Affairs, 4 (December 1952), pp. 401-4.
- Dobell, Peter C. Canada's Search for New Roles: Foreign Policy in the Trudeau Era. London: Oxford University Press for the Royal Institute of International Affairs, 1972, pp. 136-42.
- Dobell, W. M. "Canada's Record at UNCTAD III and the Meaning of Santiago." *International Perspectives*, (September-October, 1972), pp. 21-9.
- Doupe, Jack H. "Foreign Aid and Liberty." Canada Month, 7 (March 1967), p. 5.
- Dudley, Leonard; and Montmarguette, Claude. "A Model of Supply of Bilateral Foreign Aid." *The American Economic Review*, 66 (March 1976), pp. 132-42.

E

- "E.A.O. Becomes C.I.D.A." Foreign Trade, 130 (November 9, 1968), p. 12.
- Eayrs, James. The Art of the Possible: Government and Foreign Policy in Canada. Toronto: University of Toronto Press, 1961, pp. 58-61.
- Eayrs, James. Canada in World Affairs: October 1955 to June 1957. Toronto: Oxford University Press, 1959, pp. 193-213.
- "Elephants Won't Work in the Afternoon." Cooperation Canada, 2 (May-June 1972), pp. 21-6.

F

- Fatouros, Arghyrios A.; and Kelson, Robert N. Canada's Overseas Aid. Toronto: Canadian Institute of International Affairs, 1964. (Contemporary Affairs No. 31)
- Fairweather, Gordon. "Youth and Responsibility." Canadian Business, 42 (January 1969), pp. 34-40.
- "Federal and Quebec Governments Co-operate on Major Aid Project." External Affairs, 22 (April 1970), pp. 101-2.
- Fischer, Lewis A. "Canadian Wheat in a Hungry World." Behind the Headlines, 25 (April 1966).
- Fletcher, Mary. "Examining UNCTAD's Plan to Help the 'Hard-Core' 25." International Perspectives, (July-August 1972), pp. 19-23.

- Fontaine, F. Canadian Foreign Aid: Philosophy, Sociology and Policy. Ottawa: Canadian International Development Agency 1969. (CIDA Policy Review)
- Fraser, Colin. "Le rôle essentiel de la communication dans les programmes de développement." Coopération Canada, 14 (mai-juin 1974), pp. 14-19 (Eng. "Involving the People in Development Projects").
- Fraser, R. Blair. "Are We Pulling Our Weight Overseas?" *Maclean's*, 66 (January 1, 1953), pp. 4 ff.
- Fraser, R. Blair. "Backstage at Ottawa." Maclean's, 68 (October 29, 1955), pp. 6 ff.
- Fraser, R. Blair. "We're Not Doing Our Share of Foreign Aid." Maclean's, 69 (May 26, 1956), pp. 8 ff.
- "From Ottawa: CIDA In for Some Rough Grilling?" Canadian Business, 48 (March 1975), p. 7.
- Fulford, Robert. "One Canadian's Plea for a New Canadian Purpose." *Maclean's*, 75 (October 6, 1962), pp. 22 ff.

G

- Garry, Robert. "Le Tiers-Monde et nous." Relations, 37 (mai 1977), pp. 131-5.
- Gérin-Lajoie, Paul. Canadian International Cooperation: Approaches to the Seventies. Ottawa: Canadian International Development Agency, 1971, (Thoughts on International Development, No. 1)
- Gérin-Lajoie, Paul. "CIDA in a Changing Government Organization." Canadian Public Administration, 15 (Spring 1972), pp. 46-58.
- Gérin-Lajoie, Paul. "CIDA Should Become the Developing Countries' Spokesman in Canada." Ceres, 33 (May-June 1973), pp. 28-31.
- Gérin-Lajoie, Paul. "La coopération au développement international: dilemmes et choix." *Coopération Canada*, 21 (juillet-août 1975), pp. 12-21 (Eng. "International Development Cooperation: Dilemmas and Choices").
- Gérin-Lajoie, Paul. The Development Officer. Ottawa: Canadian International Development Agency, 1971. (Thoughts on International Development, No. 2)
- Gérin-Lajoie, Paul. Developmental Administration. Ottawa: Canadian International Development Agency, 1972. (Thoughts on International Development, No. 4)
- Gérin-Lajoie, Paul. Educational Innovation. Ottawa: Canadian International Development Agency, 1971. (Thoughts on International Development, No. 3)
- Gérin-Lajoie, Paul. Focus on Man. Ottawa: Canadian International Development Agency, 1974. (Thoughts on International Development, No. 7)
- Gérin-Lajoie, Paul. Journey to Justice. Ottawa: Canadian International Development Agency, 1971. (Thoughts on International Development, No. 5)
- Gérin-Lajoie, Paul. The Longest Journey ... begins with the first step. Ottawa:

- Canadian International Development Agency, 1976. (Thoughts on International Development, No. 10)
- Gérin-Lajoie, Paul. No Room for Routine. Ottawa: Canadian International Development Agency, 1972. (Thoughts on International Development, No. 6)
- Gérin-Lajoie, Paul. The Path We Have Come ... Highlights 1971 / 1976. Ottawa: Canadian International Development Agency, 1976. (Thoughts on International Development, No. 11)
- Gérin-Lajoie, Paul. "Plus d'aide, oui; mais plus que de l'aide." Coopération Canada, 17 (novembre-décembre, 1974), pp. 20-1 (Eng. "There is Not Enough Aid, and Aid is Not Enough").
- Gérin-Lajoie, Paul. "Reflections on 25 Years of Development Co-operation." International Perspectives, (November-December 1975), pp. 38-42.
- Gérin-Lajoie, Paul. "A Selective Strategy: Canadian Aid Aims at Helping Countries Working Toward Self-Reliance." *Ceres*, 9 (January February 1976), pp. 46-9.
- Gérin-Lajoie, Paul. Shaping Tomorrow. Ottawa: Canadian International Development Agency, 1975. (Thoughts on International Development, No. 9)
- Gérin-Lajoie, Paul. 25 Years After. Ottawa: Canadian International Development Agency, 1975. (Thoughts on International Development, No. 8)
- Gillis, Frank. "CUSO Workers and International Development." Canadian Labour, 20 (September 1975), pp. 15-16.
- Glyde, Henry. "What Canada Can Learn About Science Policy From Developing Nations." Science Forum, 6 (October 1973), pp. 26-9.
- "Go Overseas, Old Boy." Executive, 15 (March 1973), pp. 53-4.
- Goldfarb Consultants Ltd. Perspectives on the Canadian International Development Agency: A Research Report for the Canadian International Development Agency. Ottawa: 1976.
- Gordon, J. King (ed.). Canada's Role as a Middle Power: Papers given at the Third Annual Banff Conference on World Development, August 1965. Toronto: The Canadian Institute of International Affairs, 1966, pp. 145-66. (Contemporary Affairs, No. 35)
- Gordon, Sheldon. "Canadian Aid Policy: What's In It for Us?" International Perspectives, (May-June 1976), pp. 21-5.
- Grafftey, W. Heward. "Canada and the Third World." Canadian Business, 47 (September 1974), pp. 56-8.
- Gregory, John B. "Can CIDA Decide Research Priorities for Poor Countries?" Science Forum, 8 (April 1975), pp. 19-20.
- Grey, Rodney Y. "Canadian Aid Program is Needed Both in Canada and Overseas." Saturday Night, 63 (September 11, 1948), p. 40.

H

Halifax Conference, 2nd. Canada and the New International Economic Order.

- Halifax: Saint Mary's University, 1975.
- Harker, John. "Trade Unionism and a New International Economic Order." *International Perspectives*, (September / October 1976), pp. 11-15.
- Harrison, W. E. B. Canada in World Affairs: 1949 to 1950. Toronto: Oxford University Press, 1957, pp. 217-34.
- Helleiner, Gerald K. "Canada and the New International Economic Order." Canadian Public Policy, 2 (Summer 1976), pp. 451-65.
- Helleiner, Gerald K. "The Development Business: Next Steps." *International Journal*, 25 (Winter 1969-70), pp. 158-77.
- Heyneman, Donald. "Why We Must Prevent Foreign Aid From Becoming an Ecological Nightmare." Science Forum, 4 (October 1971), pp. 3-9.
- Higgins, Benjamin. "The Evaluation of Technical Assistance." *International Journal*, 25 (Winter 1969-70), pp. 34-55.
- Higgins, Benjamin; and Higgins, Jean. Canada's Trade Policy in the Second Development Decade. Montreal: Private Planning Association, 1970.
- Higgins, Benjamin; and Fréchette, Pierre. Six Studies. Ottawa: Canadian International Development Agency, 1969 (CIDA Policy Review).
 - (1) The Terms of Aid; (2) The Positive Foreign Aid Policy for Canadians;
 - (3) Trade not Aid?; (4) Liberalized Immigration as an Alternative to Aid;
 - (5) The Costs of Tying Aid; (6) Cost-Benefit Analysis of Trade Versus Aid.
- Hilborn, Kenneth. "Aiding the Underdeveloped." *Canada Month*, 9 (March 1969), p. 27.
- Hitschmanova, Lotta. *The USC Story*. Ottawa: Unitarian Service Committee of Canada, 1970.
- Holmes, John W. "Economic Aid in Canada's Middlepower Diplomacy Creates Goodwill Among Underdeveloped Countries." *Canadian Business*, 34 (April 1961), pp. 118-25.
- Holmes, John W. "The West and the Third World." *International Journal*, 21 (Winter 1965-66), pp. 20-41.
- Hopper, W. David. Canada's Role in World Agricultural Development. Ottawa: International Development Research Centre, 1977 (IDRC-085e).

 Also available in French (IDRC-085f).
- Hopper, W. David. Development at the Crossroads; Present Problems, Future Prospects. Ottawa: International Development Research Centre, 1977 (IDRC-090e).
- Hopper, W. David. *The Politics of Food*. Ottawa: International Development Research Centre, 1977 (IDRC-100e).
- Hopper, W. David. Research Policy: Eleven Issues. Ottawa: International Development Research Centre, 1973 (IDRC-014e).

 Also available in French (IDRC-014f).
- Hopper, W. David. Statement to the Inaugural Meeting of the Board of Governors of the International Development Research Centre. Ottawa: International

- Development Research Centre, 1970 (IDRC-002e). Also available in French (IDRC-002f).
- Hopper, W. David. To Conquer Hunger: Opportunity and Political Will. Ottawa: International Development Research Centre, 1975 (IDRC-048e).

 Also available in French (IDRC-048f).
- Huff, W. G. "Canadian Bilateral Aid: Canadian Content and Balance-of-Payments Costs." *Journal of World Trade Law*, 7 (September-October, 1973), pp. 587-97.
- Hulse, Joseph H. "La nouvelle céréale miracle." Coopération Canada, 16 (septembre-octobre 1974), pp. 18-23. (Eng. "The New 'Miracle' Grain").
- Humphreys, David. "The Politics of Food." *Maclean's*, 88 (February 1975), pp. 19-21.
- Hymer, Stephen; and Van Arkadie, Brian. "Offering Options to the 'Third World'." An Independent Foreign Policy for Canada? Edited by Stephen Clarkson. Toronto: McClelland & Stewart, 1968, pp. 226-43.

I

- Innis, Hugh (ed.). Issues for the Seventies: International Involvement. Toronto: McGraw-Hill Ryerson Ltd., 1972.
- International Conference of the I.C.I., 1st., University of Ottawa, 1968. Les forces vives du développement. Dynamics of Development. Ottawa: University of Ottawa Press, 1974. (Papers and Documents of the I.C.I. Series A: No. 1)
- International Conference of the I.C.I., 2nd., University of Ottawa, 1970. L'expert international. The International Advisor. Ottawa: University of Ottawa Press, 1974. (Papers and Documents of the I.C.I. Series A: No. 2)
- International Conference of the I.C.I., 3rd., University of Ottawa, 1971. L'action dialectique de la recherche sur les études en développement international. Research and Studies in International Development. Ottawa: University of Ottawa Press, 1974. (Papers and Documents of I.C.I. Series A: No. 3)
- "International Action to Avert a Protein Crisis." External Affairs, 21 (March 1969), pp. 140-2.
- International Development Research Centre. Agriculture, Food and Nutrition Sciences Division: The First Five Years. Ottawa: 1977. (IDRC-089e).
- International Development Research Centre. IDRC Programs in Agriculture, Fisheries, Forestry, and Food Science: Reviewed at a Symposium, Ottawa, September 12, 1973. Ottawa: 1974. (IDRC-029e).
- International Development Research Centre. Population and Health: IDRC Program Directions. Ottawa: 1975 (IDRC-046e).

 Also available in French.
- "International Development Research Centre." External Affairs, 23 (January 1971), pp. 34-6.
- Islam Nasir. "New Ethnics and Politics of World Food Scarcity." *International Perspectives*, (November / December 1976), pp. 18-22.

- Jeanes, C. W. L. "Esculape doit céder une large place aux auxiliaires médicaux." Coopération Canada, 14 (mai-juin 1974), pp. 20-2. (Eng. "Kidney Machines Versus Nutrition and Hygiene").
- Johnson, Harry G. "Pearson's 'Grand Assize' Fails: A Bleak Future for Foreign Aid." Round Table, 60 (January 1970), pp. 17-25.
- "Joint Church Project on Third World Needs." Canadian Labour, 18 (March 1973), p. 32.
- "Jungle à la moderne: la forêt tropicale, un atout maître du Tiers-Monde." *Coopération Canada*, 2 (mai-juin 1972), pp. 21-6.
- Juvet, C. S. "Training in Public Administration at Carleton University Under Canada's External Aid Program." Canadian Public Administration, 4 (1961), pp. 396-405.

K

- Kaplansky, Kalmen. "The Economic Policies of a Middle Power: Canada's Contribution to International Economic Cooperation." Canada's Role as a Middle Power: Papers given at the Third Annual Banff Conference on World Development, August 1965. Edited by J. King Gordon. Toronto: The Canadian Institute of International Affairs, 1966, pp. 145-66. (Contemporary Affairs No. 35)
- Kavic, Lorne. "Canada and the Commonwealth." *Round Table*, 65 (January 1975), pp. 37-49.
- Keenleyside, Hugh L. International Aid: A Summary With Special Reference to the Programmes of the United Nations. Toronto: McClelland and Stewart, 1966.
- Keirstead, B. S. Canada in World Affairs: September 1951 to October 1953. Toronto: Oxford University Press, 1956, pp. 205-14.
- Keyfitz, Nathan. "Foreign Aid Can Be Rational." *International Journal*, 17 (Summer 1962), pp. 237-50.
- Kidd, J. R. "A Market Place for Ideas as Well as Goods." Saturday Night, 74 (May 23, 1959), pp. 16 ff.

L

- Latin American Working Group. Canadian Aid: Whose Priorities? A Study of the Relations Between Non-Governmental Organizations, Business and the Needs of Latin America. Toronto: Latin American Working Group, 1973.
- Laird, Marshall. "A Novel Attempt to Control Biting Flies With Their Own Diseases." Science Forum, 5 (December 1972), pp. 12-14.
- Leach, Richard H. "Canada and the Commonwealth: Assuming a Leadership Role." *International Perspectives*, (May-June 1973), pp. 24-9.
- Lebel, Ronald. "Foreign Aid is Big Business." Canadian Business, 39 (September 1966), pp. 94-102.
- Leroy, Marcel. "Neo-Malthusianism, Foreign Aid and International Relations." *International Journal*, 31 (Winter 1975-6), pp. 26-43.

- "Let's Stop the Fourth World War Too." Macleans, 64 (January 15, 1951), p. 1.
- Linton, Neville. "The Hard Facts of Foreign Aid." Commentator, 12 (July-August 1968), pp. 12-14.
- Lloyd, Trevor. Canada in World Affairs, 1957-1959. Toronto: Oxford University Press, 1968, pp. 144-7; 195-201.
- Lotz, Jim. "Foreign Aid and Domestic Difficulties." *Issues for the Seventies: International Involvement*. Edited by Hugh Innis. Toronto: McGraw-Hill Ryerson Ltd., 1972, pp. 33-6.
- Lotz, Jim. "Foreign Aid and Domestic Difficulties." Canadian Forum, 49 (June 1969), pp. 54-5.
- Lucas, R. D. "United Nations' Development Program." Foreign Trade, 133 (January 31, 1970), pp. 14-18.
- Lucas, R. D. "The United Nations' Development Program: Its Implications for the Consulting Engineer." *Foreign Trade*, 130 (December 7, 1968), pp. 5-9.
- Lux, André. "Traditions, systèmes économiques et les contraintes du développement." *International Journal*, 25 (Winter 1969-70), pp. 1-8.
- Lyon, Peyton V.; and Tareq Y. Ismael (eds.). Canada and the Third World.

 Toronto: Macmillan of Canada, 1976.
- Lyon, Peyton V. Canada in World Affairs, 1961-1963. Toronto: Oxford University Press, 1968, pp. 351-9.
- Lyon, Peyton V. The Policy Question: A Critical Appraisal of Canada's Role in World Affairs. Toronto: McClelland and Stewart, 1963, pp. 32-5.

M

- McConkey, David. "Pour une éducation sans viellières, apte à formes des citizens du monde." *Coopération Canada*, 16 (septembre-octobre, 1974). (Eng. "Education for a Global Lifestyle").
- Macdonald, Donald S. "Economic Development in the Developing Countries." External Affairs, 18 (November 1966), pp. 477-81.
- MacEachen, Allan J. "Canada's Contribution to Rural Development: Special Report." Agriculture Abroad, 30 (April 1975), pp. 32-5.
- MacEachen, Allan J. General Policies of the Canadian International Development Agency, 1975. Statement made to the Standing Committee of the House of Commons on External Affairs and National Defence, April 10, 1975. Ottawa: Canadian International Development Agency, 1975.
- MacEachen, Allan J. International Development Cooperation. Statements to the Seventh Special Session of the United Nations General Assembly, September 3, 1975, and the Standing Committee of the House of Commons on External Affairs and National Defence, April 10, 1975. Ottawa: Canadian International Development Agency, Information Division, September 1975.
- MacEachen, Allan J. Report on Canada's Food Aid Programme: Allocation for 1975-76. Ottawa: Canadian International Development Agency, 1975.
- MacEachen, Allan J. Statement on the Roles of Canada and Canadian International

- Development Agency in Providing Development Assistance. Statement before the Standing Committee of the House of Commons on External Affairs and National Defence, April 10, 1975. Ottawa: Canadian International Development Agency, 1975.
- McInnis, Edgar. "Canadian Opinion and Foreign Policy." Queen's Quarterly, 62 (Winter 1955), pp. 504-14.
- MacKenzie, A. E. D. "Le Canada et 'le nouvel ordre économique'." *Coopération Canada*, 21 (juillet-août 1975), pp. 3-11. (Eng. "Canada and the New Economic Order").
- McKinnell, Robert. "External Aid: How to Choose." An Independent Foreign Policy for Canada? Edited by Stephen Clarkson. Toronto: McClelland and Stewart, 1968, pp. 244-52.
- MacLean, Logan. "Ottawa Letter." Saturday Night, 73 (October 11, 1958), p. 6.
- McLoughlin, Peter F. M. "Rationalizing Foreign Aid." *International Journal*, 18 (Winter 1962-63), pp. 87-90.
- McWhinney, Bill; and Godfrey, Dave (eds.). Man Deserves Man: CUSO in Developing Countries. Toronto: Ryerson Press, 1968.
- Magee, David. "CESO Living in a World the Tourists Never See." Canada Commerce, 137 (March 1973), pp. 25-6.
- Maione, Roméo. "Les ONG et le développement." Coopération Canada, 5 (novembre-décembre 1972), pp. 20-23 (Eng. "A Bridge of Understanding and Action: An NGO View of Development").
- Malhotra, S. P. Canadian Mineral Aid and Private Investment in the Developing Countries. Ottawa: Information Canada, 1974. (Mineral Bulletin MR143)
- Martin, Paul. Canada and the Quest for Peace. New York: Columbia University Press, 1967, pp. 66-93. (Jacob Blaustein Lectures on International Affairs).
- Martin, Paul. "Canada's Aid Programs and the Business Community." *Foreign Trade*, 126 (November 26, 1966), pp. 27-31.
- Martin, Paul. "Canada's Foreign Relations." Ottawa: Department of External Affairs, 1971.
- Martin, Paul. "Canadian Development Assistance." External Affairs, 19 (July 1967), pp. 285-96. (Jacob Blaustein Lectures for 1967, Columbia University, April 28, 1967).
- Martin, Paul. "Meeting of Development Assistance Committee." External Affairs, 18 (September 1966), pp. 377-82.
- Martin, Paul. "Expansion of Assistance Programme; Statement in the House of Commons." External Affairs, 15 (December 1963), pp. 470-73.
- Martin, Paul. Paul Martin Speaks for Canada: A Selection of Speeches on Foreign Policy, 1964-1967. Toronto: McClelland and Stewart, 1967, pp. 129-38.
- Martin, Paul. "Principles and Purposes of Foreign Aid." External Affairs, 17 (March 1965), pp. 89-95.
- Martin, Paul. Speech to the Knights of Columbus, Sarnia, Ontario, June 8, 1974.

- Ottawa: Department of External Affairs, 1974.
- Martin, Paul. "Why We Invest in Foreign Aid." Canadian Business, 38 (May 1965), pp. 46 ff.
- Martin, Robert. "Who Suffers Whom? Notes on a Canadian Policy Towards the Third World." Canadian Forum, 56 (April 1976), pp. 12-19.
- Mason, Edward S.; and Asher, Robert E. The World Bank Since Bretton Woods. Washington, D.C.: Brookings Institution, 1973, pp. 19 ff.
- Masters, Donald C. Canada in World Affairs: 1953 to 1955. Toronto: Oxford University Press, 1959, pp. 124-9.
- Matthews, Roy A.; and Proudfoot, Stuart B. Ways of Increasing the Involvement of Canadian Private Interests in the Developing Countries. A Report of the Private Planning Association to the Canadian International Development Agency. Ottawa: Canadian International Development Agency, 1969. (CIDA Policy Review)
- Mboya, Tom J. "Technical Assistance." Canadian Forum, 68 (October 1968), pp. 147-8.
- Meagher, M. "United Nations Conference on Trade and Development." External Affairs, 17 (June 1965), pp. 225-8.
- "Meanwhile, in the Homefront." Executive, 15 (March 1973), pp. 54.
- "Medical Aid from Canada." External Affairs, 18 (May 1966), p. 217.
- Mende, Tibor. "Vers une 'révision déchirante' du développement et de 'l'aide'." Coopération Canada, 6 (janvier-février 1973), pp. 3-9 (Eng. "The Development Crisis: The Real Questions").
- Meredith, Brian. "What Follows the UN Food Conference?" Canadian Business, 48 (March 1975), pp. 23-6.
- Mills, J. C. "Canada at UNCTAD." *International Journal*, 20 (Spring 1965), pp. 214-20.
- Mills, Joseph C. "The United Nations Conference on Trade and Development." Canadian Banker, 72 (Summer 1965), pp. 57-66.
- "Mr. Diefenbaker Errs." Saturday Night, 74 (August 1, 1959), p. 4.
- Moran, H. O. "Canada's Bilateral Grant Aid Programmes." External Affairs, 16 (February 1964), pp. 67-84; (September 1964), pp. 423-40.
- Moran, H. O. "The Problems of Under-development." External Affairs, 18 (June 1966), pp. 245-8.
- Morisette, Brigitte. "Le French Power à Ottawa: Paul Gérin-Lajoie." Le Magazine Maclean, 11 (juin 1971), pp. 42-5.
- Morris, Robert C. Overseas Volunteer Programs: Their Evolution and the Role of Governments in Their Support. Lexington, Mass.: Lexington Books, 1973, pp. 61-81.

N

Nash, C. Knowlton. "Why Don't We Send Our Surplus Food to the Starving?"

- Maclean's, 72 (December 19, 1959), pp. 21 ff.
- National Committee on Canada's Participation in Social Development Abroad. Canada's Participation in Social Development Abroad. Ottawa: September 1963. (Carleton University Workshop, June 16-19, 1963)
- "New Aid Program: Canadian Sales Probably Not Affected." Canada Commerce, 139 (December 1975), pp. 10-1.
- "Nonprofit Organizations Aid Development Abroad." Canadian Welfare, 50 (January-February 1974), pp. 22-3.
- North-South Institute. North South Encounter; The Third World and Canadian Performance. Ottawa: 1977. (Canada North South, 1977-1978, V.1)
- North-South Institute. Third World Deficits and the 'Debt Crisis'. Ottawa: 1977. (Canada North South, 1977-1978, V.2)

0

- Ohlin, Goran. Foreign Aid Policies Reconsidered. Paris: Development Centre of the Organisation for Economic Co-operation and Development, 1966, p. 48. (Development Centre Studies)
- Organisation for Economic Co-operation and Development. *Aid Evaluation*. Paris: 1975, pp. 21-4.
- Organisation for Economic Co-operation and Development. *The Flow of Financial Resources to Less-Developed Countries 1956-1963*. Paris: 1964, pp. 69-71.
- Organisation for Economic Co-operation and Development. *The Flow of Financial Resources to Less-Developed Countries 1961-1965*. Paris: 1967, pp. 71-2.
- Organisation for Economic Co-operation and Development. *The Flow of Financial Resources to Less-Developed Countries 1961-1971*. Paris: 1973, pp. 67-88.
- Organisation for Economic Co-operation and Development. Geographical Distribution of Financial Flows to Developing Countries. Data on Disbursements 1969-1975. Paris: 1977, pp. 262-83.
- Organisation for Economic Co-operation and Development. Resources for the Developing World: The Flow of Financial Resources to Less-Developed Countries 1962-1968. Paris: 1970, pp. 47-58.
- "Our Good Grey Record Abroad." Maclean's, 79 (November 5, 1966), p. 4.
- Overseas Institute of Canada. Canada's Participation in Overseas Development.

 A Report of the Second National Workshop on Canada's Participation in International Development, Esterel, P.Q. November 18-21, 1965. Ottawa: n.d.

P

- Pal, D. D. (ed.). Canadian Economic Issues: Introductory Readings. Toronto: Macmillan of Canada, 1971, Part X: External Aid, pp. 570-610.
- "Palestine Relief Operation Crisis." External Affairs, 43 (January 1971), pp. 41-2.
- "Participez à la croissance de pays en voie de développement." Commerce, 73 (janvier 1971), p. 44.

- Pearson, Lester B. *The Crisis of Development*. New York: Praeger Publishers for the Council on Foreign Relations, 1970. (Russell C. Leffingwell Lectures, 1969)
- Pearson, Lester B. Mike: Volume II 1948-1957. Toronto: University of Toronto Press, 1973, pp. 116-23.
- Pearson, Lester B. Partners in Development: Report of the Commission on International Development. New York, N.Y.: Praeger Publishers, 1969.
- Pelletier, Irénée. La Politique Canadienne d'Aide au Développement. Paris: Librairie Générale de Droit et de Jurisprudence, 1971.
- Peters, Stuart. "A Canadian Search for New Development Imperatives." *The Imperatives for Development*. Proceedings of the Seventh Annual Banff Conference on World Affairs, Banff, August 18-23, 1969. Toronto: Canadian Institute of International Affairs, 1969, pp. 166-77.
- Piché, Raymond. "The Structure of Canadian Aid." *International Development*, 2 (September 1968), p. 4; (November 1968), p. ; (June 1969), p. 7.
- Plumptre, A. F. W. "The Atlantic Countries and Aid to the Third World." *Canada and Europe: A New Look*. Proceedings of the Fifth Annual Banff Conference on World Affairs, Banff, August 21-26, 1967. n.p.n.p. 1968, pp. 108-28.
- Plumptre, A. F. W. "The International Development Research Centre and the Role of L. B. Pearson." *Freedom and Change: Essays in Honour of Lester B. Pearson*. Edited by Michael G. Fry. Toronto: McClelland and Stewart, 1975, pp. 152-170.
- Plumptre, A. F. W. "Perspective on Our Aid to Others." *International Journal*, 22 (Summer 1967), pp. 484-99.
- Plumptre, A. F. W. Three Decades of Decision; Canada and the World Monetary System, 1944-1975. Toronto: McClelland and Stewart, 1977, pp. 62-3, 72-81, 112-22.
- Pogson, T. W. S. "Foreign Aid: Key to Future Trade." Saturday Night, 77 (January 6, 1962), pp. 14-5.
- Preston, Richard A. Canada in World Affairs, 1959 to 1961. Toronto: Oxford University Press, 1965, pp. 216-36.
- "Progress in Technical Cooperation." External Affairs, 4 (December 1952), pp. 398-400.

R

- Randall, Stephen J. "A New Look at Foreign Aid." *The Empire Clubs of Canada: Addresses*, 1965-1966. Toronto: T.H. Best Printing Co. Ltd., pp. 319-32.
- Reid, Escott. "A Blueprint for Assisting the World's Poorest People." *International Perspectives* (September / October, 1977), pp. 3-7.
- Reid, Escott. "Canada and the Struggle Against World Poverty." *International Journal*, 25 (Winter 1969-70), pp. 142-57.
- Reid, Escott. "Canadian Foreign Policy, 1967-1977: A Second Golden Decade?" International Journal, 22 (Spring 1967), pp. 171-81.

- "Resource Surveys A Canadian Specialty." External Affairs, 18 (August 1966), pp. 338-9.
- Reuber, Grant L. "Canada's Economic Policies Toward the Less Developed Countries." The Canadian Journal of Economics, 1 (November 1968), pp. 669-98.
- Reuber, Grant L. Canada's Foreign Aid Policy; the Strategy and Tactics for Allocating Canadian Funds to Foster Economic Development. Ottawa: Canadian International Development Agency, 1969. (CIDA Policy Review)
- Reuber, Grant L. Canada's Interest in the Trade Problems of Less-Developed Countries. Montreal: Private Planning Association, 1964.
- Reuber, Grant L. "The Trade-Offs Among The Objectives of Canadian Foreign Aid." *International Journal*, 25 (Winter 1969-70), pp. 129-41.
- Reuber, Grant L. "Why Canadian Foreign Aid." *International Journal*, 14 (Winter 1958-59), pp. 11-20.
- Richardson, Ken. "International Financing Series: Lots of Preparation Goes Into a CIDA Project." Canada Commerce, 137 (March 1973), pp. 31-3.
- Robb, William. "New Face on External Aid." Canadian Business, 41 (May 1968), pp. 12-3.
- Robinson, H. B. "Prolongements monétaires et commerciaux de la coopération canadienne." Coopération Canada, 24 (1976), pp. 20-4. (Eng. "Trade, Monetary Matters and the Canadian Aid Program").
- Roblin, Duff; and Holmes, John. "Canada's Role." Diplomacy in Evolution. Edited by D. L. B. Hamlin. (Proceedings of the 30th Annual Couchiching Conference). Toronto: University of Toronto Press for the Canadian Institute on Public Affairs, 1961, pp. 112-24.
- Roche, Douglas. "Can Canada Help Achieve a New World Order?" Saturday Night, 90 (May 1975), pp. 11-5.
- Roche, Douglas. Justice Not Charity: A New Global Ethic for Canada. Toronto: McClelland and Stewart, 1976.
- "Rôle de l'ACDI (face au problème de la population mondiale)." Coopération Canada, 12 (janvier-février 1974), pp. 38-9 (Eng. "CIDA's Role (in dealing with the world population problem)").
- "Rôle du CRDI (face au problème de la population mondiale)." Coopération Canada, 12 (janvier-février 1974), pp. 40-41 (Eng. "The IDRC's role (in dealing with the world population problem)").
- Rubin, Seymour J. The Conscience of the Rich Nations: The Development Assistance Committee and the Common Aid Effort. New York: Harper and Row for the Council on Foreign Relations, 1966.
- Ruddick, Valley. "Foreign Aid Breakdown: A New Threat to World Order?" Science Forum, 9 (February 1976), pp. 18-9.

S

Sabourin, Louis. Analyse des politiques de coopération internationale du Canada:

- des projets d'aide à la stratégie de développement. Ottawa: Université d'Ottawa, 1975.
- Sabourin, Louis. Contrôle parlementaire et politique canadienne de développement international; une analyse de cas: examen de l'ACDI par le Comité permanent des Affairs Extérieures et de la Défense Nationale. Ottawa: Université d'Ottawa, 1975.
- Sabourin, Louis. Multilateral Aid and the Coordination of Aid. Ottawa: Canadian International Development Agency, 1969. (CIDA Policy Review)
- Sabourin, Louis. "Normes juridiques canadiennes en matière de développement international." Canadian Perspectives on International Law and Organization. Edited by R. St. J. Macdonald. Toronto: University of Toronto Press, 1974, pp. 794-813.
- Sabourin, Louis. "Ottawa et Québec dans l'Agence: une coopération à inventer." Perspectives internationales, (janvier-février 1972), pp. 18-24.
- Sanger, Clyde. "Canada and Development in the Third World." Canada and the Third World. Edited by P. V. Lyon and T. Y. Ismael. Toronto: Macmillan of Canada, 1976, pp. 277-306.
- Sanger, Clyde. "Environment and Development." *International Journal*, 28 (Winter 1972-73), pp. 103-120.
- Sanger, Clyde. Half a Loaf: Canada's Semi-Role Among Developing Countries.
 Toronto: Ryerson Press, 1969.
- Sanger, Clyde. "The International Development Research Centre: Canada's Help to the World's Rural Poor." Canadian Geographical Journal, 88 (May 1974), pp. 14-21.
- Sarbadhikari, P.; and Jecchinis, C. "The Nature of Canadian International Development Aid." Revue iranienne de relations internationales, No. 9 (printemps / été 1977), pp. 143-56.
- Schiele, Robin. "Wanted! Executives to Work Abroad." Canadian Business, 41 (April 1968), pp. 69-70.
- Scott, F. R. "Technical Assistance and Economic Aid Through the United Nations." *Proceedings and Transactions of the Royal Society of Canada*, Third Series, 47 (June 1953), pp. 17-31.
- Scott, F. R.; MacKay, R. A.; and Ritchie, A. E. *The World War Against Poverty*. Toronto: The University of Toronto Press for the Royal Society of Canada, 1953.
- "The Second United Nations Development Decade." External Affairs, 22 (September 1970), pp. 315-9.
- Sharp, Mitchell. "An International Development Research Centre for Canada." External Affairs, 22 (February 1970), pp. 45-9.
- "Should We Bring Foreign Aid Back Home?" Maclean's, 69 (April 28, 1956), p. 2.
- "Siblin Vocational Training Centre." External Affairs, 15 (January 1963), pp. 37-8.
- Sir George Williams University Annual Seminar on International Affairs, 5th,

- November 1963. Aspects of International Aid. Montreal: The Publication's Board, November 1963.
- Soward, F. H. Canada in World Affairs, From Normandy to Paris, 1944-1946.
 Toronto: Oxford University Press, 1950, pp. 81-123, 168-97.
- Spencer, Robert A. Canada in World Affairs, From UN to NATO, 1946-1949.
 Toronto: Oxford University Press, 1959, 153-206.
- "Sphere of Influence in the Sun." Maclean's, 80 (February 1967), p. 4.
- Spicer, Keith. "Quelques problèmes des experts techniques canadiens à l'étranger." Canadian Public Administration, 5 (1962), pp. 209-18.
- Spicer, Keith. A Samaritan State? External Aid in Canada's Foreign Policy.
 Toronto: University of Toronto Press, 1966.
- Spurgeon, David. "IDRC: Pioneering a New Style of International Aid Agency." *International Perspectives*, (May / June 1972), pp. 41-4.
- Spurgeon, David. "A New Approach to Foreign Aid: The IDRC of Canada." Science and Public Affairs Bulletin of the Atomic Scientists, 28 (November 1972), pp. 33-6.
- Stabler, Ernest. "The University, International Education, and Development Assistance." Journal of Educational Thought, 4 (August 1970), pp. 82-93.
- Stein, David Lewis. "Canada Runs 10th in Foreign Aid, a U.S. Critic Charges." Maclean's, 75 (July 14, 1962), pp. 2-3.
- "Stratégie 1975-1980: un nouveau départ de la coopération canadienne au développement." Coopération Canada, 22 (septembre-octobre 1975), pp. 3-16. (Eng. "A New Departure for Development Cooperation").
- Strong, Maurice. Canada's Assistance to Developing Nations. Vienna: Vienna Institute for Development, 1969.
- Strong, Maurice. Canada's Assistance to Developing Nations. (Inaugural lecture given at the Institute for International Cooperation, Ottawa, March 5, 1970). Ottawa: Institute for International Cooperation, 1970. (Papers and Documents of the I.C.I., Series D: No. 4, Special Conferences)
- Strong, Maurice. "The Canadian Aid Program." Canadian Forum, 48 (October 1968), pp. 156-7.
- Strong Maurice. "International Development Canada's Centennial Challenge." The Empire Club of Canada: Addresses, 1966-1967. Toronto: T.H. Best Printing Co. Ltd., pp. 172-86.
- Strong, Maurice. "North America and the Developing Nations." Canada and the United States in the World of the Seventies. Edited by R. H. Wagenberg. Windsor: University of Windsor Press, 1970, pp. 80-90.
- Sutherland-Brown, Malcolm. "The Line of Credit." Canada Commerce, 138 (February 1974), pp. 24-6.

T

Taylor, Rupert J. "Canadian Aid: A Helping Hand." Canada and the World, 42 (January 1977), p. 22.

- Thapa, B. B. The Way Between. Ottawa: International Development Research Centre, 1973. (IDRC-011e).
- Thomson, Dale C.; and Swanson, Roger F. Canadian Foreign Policy: Options and Perspectives. Toronto: McGraw-Hill Ryerson Limited, 1971, pp. 77-105.
- Triantis, Stephen G. "Backward Lands The Other Front." Behind the Headlines, 12 (February 1952).
- Triantis, Stephen G. "Canada's Interest in Foreign Aid." World Politics, 24 (October 1971), pp. 1-18 (slightly modified version); also Ottawa: Canadian International Development Agency, 1969. (CIDA Policy Review).
- Triantis, Stephen G. Foreign Aid: A New Canadian Experience. Texts of Addresses delivered at the 25th Annual Couchiching Conference, August 4th-11th, 1956. Toronto: Canadian Institute on Public Affairs, 1956, pp. 55-63.
- Triantis, Stephen G. The Place of Foreign Aid in Development Strategy. Ottawa: Carleton University, School of International Affairs, 1968. (Occasional Paper No. 4).
- "The Trouble with Foreign Aid." Time, 105 (June 2, 1975), pp. 8-9.
- Trudeau, Pierre Elliott. "International Development as a Requisite for Peace." External Affairs, 20 (June 1968), pp. 246-50.
- Trudeau, Pierre Elliot. "Tied Aid or Not?" Ceres, 1 (September-October 1968), p. 14.

U

- "UN Relief for Palestine Refugees." External Affairs, 13 (September 1961), pp. 303-5.
- United Nations Association in Canada. Canada and International Development Programmes. Proceedings of a National Conference at the University of Toronto, June 1 and 2, 1956. Ottawa: 1956.
- United Nations Association in Canada. Canada and the United Nations in a Changing World. Report of a Conference held in Winnipeg, Manitoba, May 12-14, 1977. Ottawa: 1977, pp. 60-88.
- United Nations Association in Canada. Conference on Canadian Aid to Underdeveloped Countries. Proceedings of a National Conference in Ottawa, May 27-28, 1955. Ottawa: 1955.
- United Nations Association in Canada, Policy Committee. The Dilemmas of Development. Toronto: 1966.
- "United Nations Development Programme: Canada's Contribution for 1968." External Affairs, 19 (November 1967), pp. 478-9.
- "UNRWA Training Centre at Homs, Syria." External Affairs, 16 (May 1964), pp. 212-3.
- Usher, Dan. The Level of Aid. Ottawa: Canadian International Development Agency, 1969. (CIDA Policy Review)

V

Virtue, David. "Something to Think About: For Many Death is a Way of Life." Canada Commerce, 139 (October 1975), pp. 44-5.

W

- Wagenberg, R. H. (ed.). Canada and the United States in the World of the Seventies. (Proceedings of the 9th Annual University of Windsor Seminar on Canadian American Relations, 1967) Windsor: University of Windsor Press, 1970, pp. 75-90.
- Walmsley, Norma. Canadian Universities and International Development. A Report prepared for the Canadian International Development Agency. Ottawa: Association of Universities and Colleges in Canada, 1970.
- Westall, Stanley. "Canada's Role in the Aid Program." Series of Articles in *The Toronto Globe and Mail*. (February 18, 19, 20, 21, 22, 25, and 26, 1963), p. 7.
- White, Charles A. "Foreign Aid: Where Canada Stands." Canada and the World, 38 (March 1973), pp. 5-6.
- Williams, Bruce M. "How Canada Cooperates with Developing Countries." *Review of International Affairs*, 26 (October 20, 1975), pp. 28-32.
- Williams, Peter. Aid in the Commonwealth. London: Overseas Development Institute, 1965, pp. 17-8.
- Wright, Hume. "Problems of International Aid to Underdeveloped Areas." *International Journal*, 7 (Autumn 1952), pp. 265-73.

Z

Zagorin, Ruth K. *The Sociology of Food*. Ottawa: International Development Research Centre, 1977. (IDRC-099e).

Chapter 2

Regional Aid Programs

2.1 Asia

- Adams, David G. "CIDA Opens Up Markets in India." Canada Commerce, 136 (February 1972), p. 48.
- "Aid for Asia." Saturday Night, 66 (November 7, 1950), p. 2 f.
- "Aid to Burma." External Affairs, 20 (July 1968), pp. 304-5.
- "Aid to Indonesia and Malaysia." External Affairs, 16 (October 1964), pp. 502-3.
- Anderson, Bob; and Mezei, Kathy. "Welcome to the Party: Aid for Bangladesh?" Canadian Forum, 55 (June 1975), pp. 20-3.
- Angus, H. F. "Canada and Colombo: Political Aims & Effects." Queen's Quarterly, 61 (Autumn 1954), pp. 337-44.
- Angus, H. F. Canada and the Far East: 1940-1953. Toronto: University of Toronto Press, 1953, pp. 77-93.
- "Arms for India." External Affairs, 15 (June 1963), p. 250-1.
- "Asian Development Bank." External Affairs, 18 (February 1966), pp. 65-7.
- "The Asian Development Bank." External Affairs, 19 (March 1967), pp. 80-5.
- Belshaw, Cyril S. "Canadian Policy and Asian Society." *International Journal*, 11 (Winter 1955-56), pp. 36-45.
- Bentley, C. F. Report of the Canadian Agricultural Task Force on India to the External Aid Office of Canada. Ottawa: External Aid Office, 1968.
- Berrington, P. D. "Outlook for Canadian Private Capital in Asia." *International Journal*, 11 (Summer 1956), pp. 205-15.
- Bonin, Bernard. "La participation canadienne au plan de Colombo." L'Actualité Economique, 34 (octobre-décembre 1958), pp. 400-13.
- Brecher, Irving; and Abbas, S. A. Foreign Aid and Industrial Development in Pakistan. Cambridge: Cambridge University Press, 1972, pp. 72-4 ff. (Perspectives in Development)
- Bryce, Murray D. A Programme for Canadian Aid to Ceylon. Prepared for the External Aid Office, Canada. Vancouver: Canadian Projects Ltd., 1968.
- "Canada-India Atomic Reactor Project." External Affairs, 8 (May 1956), pp. 113-6.
- "Canada-Vietnam: L'aide médicale." Sept-Jours, 2 (31 mars-6 avril 1968), p. 13.

- "Canada and Asia." Far Eastern Economic Review, 78 (December 16, 1972), 22 page special section.
- "Canada and Asia '74." Far Eastern Economic Review, 86 (December 27, 1974), 30 page special section.
- "Canada and Asia '75." Far Eastern Economic Review, 90 (December 26, 1975), 22 page special section.
- "Canada and Cambodia." External Affairs, 20 (December 1968), pp. 517-21.
- "Canada and Ceylon." External Affairs, 18 (March 1966), pp. 94-100.
- "Canada and Colombo." Queen's Quarterly, 61 (Autumn 1954), pp. 319-44.
- "Canada and the Colombo Plan." External Affairs, 9 (June-July 1957), pp. 206-14.
- "Canada and the Colombo Plan." External Affairs, 11 (April 1959), pp. 191-2.
- "Canada and the Colombo Plan." External Affairs, 13 (November 1961), pp. 383-93.
- "Canada and the Colombo Plan: The Umtru Project." External Affairs, 9 (August 1957), pp. 241-3.
- "Canada and the Colombo Plan: The Warsak Project." External Affairs, 11 (June 1959), pp. 125-7.
- "Canada at Katunayake." External Affairs, 17 (January 1965), pp. 33-4.
- Canada, Department of External Affairs. Canada and the Colombo Plan. Ottawa: September 1959. (Reference Papers No. 103)
- Canada, Department of External Affairs. Canada and the Colombo Plan, 1951-1961. Rev. ed. Ottawa: Queen's Printer, 1962.
- Canada, Parliament, Senate, Standing Committee on Foreign Affairs. *Proceedings* ... Respecting the Pacific Area. 28th Parliament, 3rd. Session, 1970-71, Nos. 1-17; 4th Session, 1971, No. 1 (Final Report of the Committee Respecting Canadian Relations with the Countries of the Pacific Region.) Ottawa: 1970-71.
- "Canada's Industry Minister Visits South Asia." External Affairs, 19 (June 1967), pp. 203-5.
- "Canada's Relations with Burma." External Affairs, 19 (May 1967), pp. 171-3.
- "Canada's Relations with Thailand." External Affairs, 19 (July 1967), pp. 268-70.
- "Canadian Aid for the Aborigines of Malaysia." External Affairs, 21 (January 1969), pp. 33-5.
- "Canadian Gift to Vietnamese Children." External Affairs, 21 (January 1969), pp. 44-5
- Canadian International Development Agency. Canada's Development Assistance to Asia. Ottawa: 1974. (Briefing Papers)
- Canadian International Development Agency. Colombo Plan: Background and Summary. Ottawa: October, 1969.

- "Canadian Military Assistance to Malaysia." External Affairs, 17 (April 1965), pp. 139-40.
- Carr-Gregg, John Ralph Edward. *The Colombo Plan: A Commonwealth Program* for Southeast Asia. New York: Carnegie Endowment for International Peace, 1951. (International Conciliation, No. 467).
- Cavell, R. G. Nik. "Canada and Colombo: What We Have Done." Queen's Quarterly, 61 (Autumn 1954), pp. 319-28.
- Cavell, R. G. Nik. "Canada and the Colombo Plan." Food for Thought, 12 (January 1952), pp. 6-12.
- Cavell, R. G. Nik. "Canada and the Colombo Plan." *The Empire Club of Canada: Addresses 1952-1953*. Toronto: T.H. Best Printing Co. Ltd., 1953, pp. 112-28.
- Cavell, R. G. Nik. "The Colombo Plan." Canadian Geographic Journal, 44 (May 1952), pp. 202-13.
- "Ceylon Bound." Saturday Night, 66 (September 4, 1951), p. 2.
- Chaplin, C. E. "How Canada is Helping the Fishermen of Ceylon." Canadian Geographical Journal, 61 (July 1960), pp. 22-9.
- "CIDA President Tours Asian Nations." External Affairs, 23 (July 1971), pp. 249-51.
- Clarkson, Stephen. "Just What Is Canada Accomplishing in India?" Saturday Night, 87 (October 1972), pp. 15-17.
- Cohen, Maxwell. "Nehru the Titan and His India." Saturday Night, 74 (July 18, 1959), pp. 10 ff.
- "Colombo Again." Saturday Night, 66 (February 6, 1951), p. 2.
- "The Colombo Plan." External Affairs, 4 (June 1952), pp. 208-11.
- "The Colombo Plan." External Affairs, 5 (April 1953), pp. 102-10; 5 (May 1953), pp. 138-47.
- "The Colombo Plan." External Affairs, 6 (September 1954), pp. 273-85.
- "The Colombo Plan." External Affairs, 7 (November 1955), pp. 283-8.
- Colombo Plan Bureau. The Colombo Plan 1951 / 1971 20th Anniversary. Colombo: July 1971.
- "The Colombo Plan Comes of Age." External Affairs, 20 (March 1968), pp. 137-144.
- "Colombo Plan Consultative Committee: Manila Meeting, 1971." External Affairs, 23 (May 1971), pp. 182-5.
- "Colombo Plan Council for Technical Cooperation." External Affairs, 8 (December 1956), pp. 400-4.
- "Colombo Plan Projects in India and Pakistan." External Affairs, 13 (March 1961), pp. 74-7.

- "The Colombo Plan: A Progress Report." External Affairs, 8 (June 1956), pp. 164-9.
- "Colombo Plan Recent Developments." External Affairs, 3 (September 1951), pp. 313-4.
- "Colombo Plan Review." External Affairs, 17 (October 1965), pp. 439-44.
- "Consortia in Aid of India and Pakistan." External Affairs, 13 (August 1961), pp. 274-7.
- "Consultative Committee of the Colombo Plan." External Affairs, 21 (December 1969), pp. 438-45.
- Couchiching Conference of the Canadian Institute on Public Affairs, 28th, 1959. Changing Asia. Toronto: Canadian Institute on Public Affairs, 1959, pp. 25-30; 76-85.
- Culhane, Claire. "Behead and Cure: The Truth Behind Canada's Medical Aid to Vietnam." Canadian Dimension, 5 (December-January 1968-69), pp. 19-22.
- Culhane, Claire. Why is Canada in Vietnam? The Truth About Our Foreign Aid. Toronto: N.C. Press, 1972.
- Dacre, Douglas. "He's Making a Liar Out of Kipling." *Maclean's*, 67 (May 1, 1954), pp. 21 ff.
- DeWolf, Kenneth G. "The Canadian Businessman and Aid to India." Foreign Trade, 129 (March 30, 1968), pp. 2-6.
- Diefenbaker, John G. "Consortia on Aid to India and Pakistan." External Affairs, 13 (July 1961), pp. 262-3.
- Diefenbaker, John G. "Ten Years of the Colombo Plan." External Affairs, 12 (January 1960), pp. 454-5.
- Dupuis, Michel. "La médecine au Vietnam: le Canada y joue un rôle." Sept-Jours, 2 (21-27 janvier 1968), pp. 21-3; 2 (28 janvier-3 février 1968), pp. 24-6.
- Elman, Russell. Asian Experiment. Toronto: Ryerson Press, 1961.
- "External Affairs in Parliament Colombo Plan Aid to Pakistan Succor for Iran Disaster Victims Economic Assistance for Africa." External Affairs, 12 (May 1960), pp. 633-4.
- "External Affairs in Parliament: Sharing the Indus Waters." External Affairs, 12 (April 1960), p. 588.
- "External Affairs in Parliament: Speech From the Throne Colombo Plan Aid to India." External Affairs, 12 (March 1960), p. 561.
- Eyton, A. T. "India: Where the Aid Action Is." Canada Commerce, 137 (May 1973), pp. 24-7.
- Fraser, R. Blair. "Asia." Maclean's, 64 (February 15, 1951), pp. 8, 39-41.
- Fraser, R. Blair. "Backstage at Ottawa." Maclean's, 66 (November 1, 1953), pp. 77-8.
- Gérin-Lajoie, Paul. Report to the Secretary of State for External Affairs on the

- Mission for Bangladesh. Ottawa: Canadian International Development Agency, April 10, 1975.
- "Help for India." Saturday Night, 66 (March 27, 1951), p. 2.
- "How Much is Democracy in Asia Worth?" Maclean's, 67 (February 1, 1954), p. 2.
- "The Indus Waters Settlement." External Affairs, 12 (November 1960), pp. 847-52.
- Irwin, John. "Canada's Far East Stake India and Colombo: Hopes and Hazards." Saturday Night, 68 (January 24, 1953), pp. 15 ff.
- "Karachi Nuclear Power Project." External Affairs, 18 (February 1966), pp. 68-70.
- Kavic, Lorne. "Canada and the Pacific: Prospects and Challenges." Behind the Headlines, 29 (May 1970), p. 9.
- "Keeping Respectable." Saturday Night, 66 (March 6, 1951), p. 6.
- Keyfitz, Nathan. "Canada and the Colombo Plan." Behind the Headlines, 20 (February, 1961).
- Kingsmill, C. G. "The Warsak Hydro-Electric and Irrigation Project." *The Engineering Journal*, 43 (November 1960), pp. 43-52.
- "Kundah Power Development." External Affairs, 8 (July 1956), pp. 182-5.
- Lower, J. Arthur. Canada on the Pacific Rim. Toronto: McGraw-Hill Ryerson Ltd., 1975, pp. 208-16.
- Ma, Nancy. "Canada's Aid Programme: Welcome Change in Emphasis." Far Eastern Economic Review, 74 (December 18, 1971), pp. 37-8.
- MacKay, R. A.; and Ritchie, A. E. "The Colombo Plan: Two Years' Experience." The World War Against Poverty. Toronto: University of Toronto Press for the Royal Society of Canada, 1953.
- MacKay, R. A.; and Ritchie, A. E. "The Colombo Plan: Two Years' Experience." *Proceedings and Transactions of the Royal Society of Canada*, 3rd Series, 47 (June 1953), pp. 33-48.
- McLean, Ross. Canada and Asia: A Survey on Canadian Resources for Participation in the UNESCO Major Project on the Mutual Appreciation of Eastern and Western Cultural Values. Ottawa: Canadian National Commission for UNESCO, 1959.
- "Maple Leaf Cement Plant." External Affairs, 8 (December 1956), pp. 410-2.
- Marsh, Reid. "How to Solve Hong Kong's Problem." Canadian Commentator, 6 (September 1962), pp. 19 ff.
- Martin, Paul. "Visit to Asia." External Affairs, 9 (February 1957), pp. 62-73.
- "Mayurakshi Project." External Affairs, 8 (January 1956), pp. 8-14.
- "Medical Mission to Far East." External Affairs, 9 (May 1957), pp. 162-5.
- Mehling, Jean. "Le sous-développement, le plan de Colombo, et la contribution canadienne." L'Actualité Economique, 32 (janvier-mars 1957), pp. 604-31.
- "Mekong River Survey." External Affairs, 11 (December 1959), pp. 396-400.

- "Military Assistance to India." External Affairs, 15 (August 1963), p. 310.
- Mishra, Sudhakanta. The Role of Foreign Capital in Underdeveloped Economies. Allahabad: Kitab Mahal, 1967, pp. 126-30.
- Moran, H. O. "Canada's Bilateral Grant Aid Programmes: Colombo Plan." External Affairs, 16 (February 1964), p. 69; 16 (September 1964), p. 426.
- Morrison, Barrie M. "Canada and South Asia." Canada and the Third World. Edited by P. V. Lyon and T. Y. Ismael. Toronto: Macmillan of Canada, 1976, pp. 1-59.
- "Mr. Mayhew Travels." Saturday Night, 65 (May 23, 1950), p. 6.
- "Nam Ngum Power Project." External Affairs, 18 (June 1966), pp. 251-2.
- Parlour, Roger R. "Canada Aids and Trades with India." Foreign Trade, 126 (October 1, 1966), pp. 13-15.
- "Progress in the Colombo Plan." External Affairs, 6 (January 1954), pp. 8-15.
- Reid, Escott. "It's Time Canada Worked Boldly to Seat Red China in the United Nations, and *Really* Helped Have-Not Nations." *Maclean's*, 81 (March 1968), pp. 10 ff.
- Reid, Escott. "On China." Maclean's, 81 (March 1968), pp. 10 ff.
- Richardson, Ken. "Canadian Aid to South Asia." Canada Commerce, 138, (January 1974), pp. 35-6.
- Rothwell, B. E. "Reconstruction in South Korea: Building a Text-Book Printing Press." External Affairs, 7 (May 1955), pp. 140 ff.
- Smith, Evan. "Canada and the Colombo Plan." Canadian Banker, 67 (Spring 1960), pp. 50-60.
- Spicer, Keith. "The Administration of Canadian Colombo Plan Aid." *International Journal*, 16 (Spring 1961), pp. 169-82.
- Spicer, Keith. "Clubmanship Upstages: Canada's Twenty Years in the Colombo Plan." *International Journal*, 25 (Winter 1969-70), pp. 23-33.
- Tepper, Elliot. "Bangladesh: A Troubled State Coming to Terms With Itself." International Perspectives, (March-April 1973), pp. 12-6.
- "Thermal Power Plant for East Pakistan." External Affairs, 8 (May 1956), p. 135.
- Thomson, Dale C. "India and Canada: A Decade of Co-operation, 1947-1957." *International Studies* (New Delhi), 9 (April 1968), pp. 404-30.
- van Praagh, David. "Canada and Southeast Asia." Canada and the Third World, Edited by P. V. Lyon and T. Y. Ismael. Toronto: Macmillan of Canada, 1976, pp. 301-42.
- Warren, J. H. "Canada and Colombo: The Economics of the Plan." Queen's Quarterly, 61 (Autumn 1954), pp. 328-37.
- "We Can't Dodge This." Saturday Night, 66 (November 7, 1950), pp. 6-7.
- "Wheat: Canada's Contribution." Saturday Night, 66 (February 6, 1951), p. 6.

- "Wheat for India." Saturday Night, 66 (May 22, 1951), p. 2.
- Woodcock, George. "Aid to India Presents Some Special Problems." Saturday Night, 87 (December 1972), pp. 18-9.

2.2 Caribbean

- Berry, Glyn R. "The West Indies in Canadian External Relations: Present Trends and Future Prospects." Canadian Public Policy, 3 (Winter 1977), pp. 50-62.
- Canada, Parliament, Senate, Standing Committee on Foreign Affairs. Report on Canada-Caribbean Relations. 28th Parliament, 1st Session, 1968-69, Nos. 1-8; 2nd Session, 1969-70, Nos. 1-11; No. 12. (Final Report of the Committee Respecting Canada Caribbean Relations). Ottawa, Queen's Printer, 1969-70.
 - Chairman: The Hon. John B. Aird.
- "Canada and the Commonwealth Caribbean." External Affairs, 23 (January 1971), pp. 27-33.
- "Canadian Aid to Caribbean Countries." External Affairs, 17 (October 1965), p. 469.
- "Canadian Aid to the West Indies." External Affairs, 16 (August 1964), pp. 388-9.
- Canadian International Development Agency. Canada's Development Assistance to the Commonwealth Caribbean. Ottawa: 1974 (Briefing Papers).
- "Caribbean Development Bank Agreement." External Affairs, 22 (January 1970), pp. 26-7.
- Edmondson, Locksley G. E. "Canada and the West Indies: Trends and Prospects." *International Journal*, 19 (Spring 1964), pp. 188-201.
- Finkelstein, Rytsa. "How Canada is Helping Caribbean Development." Foreign Trade, 130 (November 23, 1968), pp. 17-8.
- Fraser, Duncan G. L. "Canada's Role in the West Indies." Behind the Headlines, 23 (January, 1964).
- "Help for Jamaica." Saturday Night, 66 (October 6, 1951), pp. 6-7.
- Leavitt, Donald H. "Canada Aids Jamaica." Foreign Trade, 133 (January 17, 1970), pp. 16-8.
- Levitt, Kari; and McIntyre, Alister. Canada-West Indies Economic Relations.

 Montreal: Private Planning Association and the Centre for Developing Area
 Studies, McGill University, 1967, pp. 107-23.
- MacQuarrie, Heath. "Canada and the Caribbean." Canada and the Third World. Edited by P. V. Lyon and T. Y. Ismael. Toronto: Macmillan of Canada, 1976, pp. 200-39.
- Midy, Franklin. "Dossier Canada-Haïti: présence canadienne et québecoise en Haïti." Relations, 381 (avril 1973), pp. 102-7.
- Moran, H. O. "Canada's Bilateral Grant Aid Programmes: Canada-West Indies Aid Programme." *External Affairs*, 16 (February 1964), p. 80; 16 (September 1964), p. 432.

- Robert, Marika. "Should We Adopt British Guiana?" Saturday Night, 79 (May 1964), pp. 19-21.
- "St. Lucia Airport Opened by Senator Martin." External Affairs, 23 (June 1971), pp. 215-7.
- "Visit of the Governor General to the Commonwealth Caribbean." External Affairs, 21 (April 1969), pp. 150-8.

2.3 Africa - General

- Africa Contemporary Record, 1968/69-. Edited by Colin Legum. New York: Africana Publishing Company, 1970-. Each volume contains information on Canada's aid allocation.
- Anglin, Douglas G. "Towards a Canadian Policy on Africa." *International Journal*, 15 (Autumn 1960), pp. 290-310.
- "Canadian Aircraft to Fight African Locusts." External Affairs, 22 (May 1970), pp. 178-80.
- Chaussé, Denise; and Chaussé, Jean-Guy. "Le projet de Babak: auto-développement à l'huile de bras." *Coopération Canada*, 23 (novembre-décembre 1975), pp. 22-7 (Eng. "The Babak Project: Do-It-Yourself Development").
- Langdon, Steven. "Canada's Role in Africa." A Foremost Nation: Canadian Policy and a Changing World. Edited by Norman Hillmer and Garth Stevenson. Toronto: McClelland and Stewart, 1977, pp. 178-201.
- Lotz, Jim. "Le développement communautaire: beaucoup de mythes et une grande vérité." *Coopération Canada*, 7 (mars-avril 1973), pp. 27-30 (Eng. "Myth and Reality in Community Development").
- Matthews, Robert. "L'Afrique noire dans la politique étrangère du Canada." Etudes Internationales, 1 (décembre 1970), pp. 59-73.
- Matthews, Robert O. "Africa in Canadian Affairs." *International Journal*, 26 (Winter 1970-71), pp. 122-150.
- Matthews, Robert O. "Canada's Relations with Africa." *International Journal*, 30 (Summer 1975), pp. 534-68.
- Porrier, Rolland P.; and Chaplin, C. E. Conference on Agricultural Research Priorities in the Economic Development of Africa: A Report. Ottawa: Department of Agriculture, 1968.
- Sanger, Clyde. "Canada's Year in Africa." Africa Contemporary Record: Annual Survey and Documents 1972-1973. Edited by Colin Legum. New York: Africana Publishing Company, 1973, pp. A81-84.
- Sanger, Clyde. "What Does Canada Care About Africa?" Africa Report, 15 (April 1970), pp. 12-5.
- Shute, J. C. M. "Canadian University Technical Assistance Programs in Africa." Canadian Journal of African Studies, 6 (1972), pp. 491-500.

2.3.1 Commonwealth Africa

- Baker, Gordon A. "The War We're Losing in East Africa." Saturday Night, 79 (November 1964), pp. 22-3.
- Bame, Kwabena N. "Canadian Contributions to African Education." Proceedings of the Seventh International Conference: Educational Strategies in the Third World, Innovations and Prospects. Ottawa: University of Ottawa Press, Institute for International Cooperation, 1976, pp. 143-7 (Papers and Documents of the I.C.I. Series A, No. 7).
- "Canada's Post-War Relief to Nigeria." External Affairs, 22 (June 1970), pp. 204-6.
- "Canada's Relations with Kenya." External Affairs, 19 (April 1967), pp. 136-8.
- "Canada's Relations with Uganda." External Affairs, 19 (June 1967), pp. 212-3.
- Canadian International Development Agency. Canada's Development Assistance to Commonwealth Africa. Ottawa: 1974 (Briefing Papers).
- Canadian International Development Agency. The Special Commonwealth Africa Assistance Plan. Ottawa: n.d.
- "Canadian Relief for Nigeria." External Affairs, 20 (August 1968), p. 320.
- Franck, Thomas. "How Canada Helped Communist China Gain a Foothold in Africa." *Maclean's*, 77 (November 2, 1964), pp. 2-3.
- "Future Canadian Aid for Tanzanian Forces." External Affairs, 17 (November 1965), pp. 495-6.
- Linton, Neville. "Canadian Work in Tanzania." Commentator, 12 (April 1968), pp. 16-8.
- Matthews, Robert J. "Canada and Anglophone Africa." Canada and the Third World. Edited by P. V. Lyon and T. Y. Ismael. Toronto: Macmillan of Canada, pp. 60-132.
- "Military Training Assistance to Ghana." External Affairs, 14 (April 1962), pp. 136-7.
- "Military Training Assistance to Nigeria." External Affairs, 15 (August 1963), pp. 295-6.
- Moran, H. O. "Canada's Bilateral Grant Aid Programmes: Special Commonwealth African Aid Programme." *External Affairs*, 16 (February 1964), p. 76; 16 (September 1964), p. 434.
- Nigeria: Canadian Aid Agreement Signal." Africa Report, 13 (February 1968), p. 40.
- "Rwanda, Land of a Thousand Hills." External Affairs, 20 (February 1968), pp. 120-1.
- "Shashi River Project in Botswana." External Affairs, 23 (February 1971), pp. 67-9.
- Steen, Jim. "Canada's Aid for Commonwealth Africa." Canada Commerce, 137 (April 1973), pp. 6-8.
- Steen, Jim. "Coopération canadienne: développement hydraulique au Ghana."

Coopération Canada, 26 (1977), pp. 11-4 (Eng. "Attacking the Water Problem in Ghana").

2.3.2 Francophone Africa

- Arsenau, Berchmans. Le Québec et l'Afrique Noire: L'Aide Québecoise et l'Afrique Francophone Sub-saharienne. Paris: Bureau d'études coopératives et communautaires, 1972.
- Berke, Robert. "Organisation progressiste des services de santé du Cameroun." Coopération Canada, 14 (mai-juin 1974), pp. 23-6 (Eng. "Public Health Program in Rural Areas Produces Results").
- "Bilan de l'assistance technique et économique canadienne au États d'Afrique."

 Revue française d'études politiques africaines, 55 (juillet 1970), pp. 45-59.
- Bland, C. L. "West and Central Africa: A Developing Market for Canadian Goods, Services and Investments." Foreign Trade, 135 (April 24, 1971), pp. 2-4.
- Canada, Department of External Affairs. Canada and 'La Francophonie'. Ottawa: February 1975 (Reference Papers, No. 137).
- "Le Canada et l'Afrique." Afrique, (1971-72), pp. 6-16. (Jeune Afrique: numéro special annuel)
- Canadian International Development Agency. Canada's Development Assistance to Francophone Africa. Ottawa: 1974. (Briefing Papers)
- Canadian International Development Agency. *Independent French-Speaking African States Aid Programme*. Ottawa: November, 1970.
- "A Canadian Mission to French-Speaking Africa." External Affairs, 20 (June 1968), pp. 274-5.
- "Canadian Relief for Morocco." External Affairs, 12 (May 1960), pp. 629-31.
- "Canadian-Tunisian Relations." External Affairs, 23 (November 1971), pp. 440-4.
- "Education Aid to French-Speaking African States." External Affairs, 13 (May 1961), pp. 186-7.
- Grenier, Raymond. "La 'crise globale' et la coopération canadienne au développement de l'Afrique francophone." *Etudes Internationales*, 5 (juin 1974), pp. 367-75.
- Henry, Jacques. "La politique canadienne d'aide à la région soudano-sahélienne." Canadian Public Policy - Analyse de Politiques, 2 (Summer - été 1976), pp. 466-81.
- Lemelin, Claude; and Marion, Jean-Claude. Le Canada français et le Tiers-Monde: L'opinion publique du Canada français et l'aide aux pays sous-développés: une enquête. Ottawa: Editions de l'Université d'Ottawa, 1963. (Cahiers des sciences sociales de l'Université d'Ottawa, 1)
- Martin, Paul. "Canada and the French Speaking World." External Affairs, 17 (October 1965), pp. 452-6.
- Martin, Paul. "Canada and 'La Francophonie'." External Affairs, 19 (April 1967), pp. 106-13.

- Meredith, Brian. "Tunisia Foreign Aid Can Pay Off." Canadian Business, 48 (July 1975), pp. 38-43.
- Moran, H. O. "Canada's Bilateral Grant Aid Programmes: Educational Programme for French-Speaking Countries in Africa." External Affairs, 16 (February 1964), p. 82; 16 (September 1964), p. 438.
- "Morocco: Canada to Aid Six Projects." Africa Report, 13 (April 1968), p. 28.
- "Mouche noire ..." Coopération Canada, 15 (juillet-août 1974), pp. 23-8 (Eng. "Black Fly ...").
- Sabourin, Louis. "Le Canada et l'Afrique francophone." Revue française d'études politiques africaines, 55 (juillet 1970), pp. 33-45.
- Sabourin, Louis. "Canada and Francophone Africa." Canada and the Third World. Edited by P. V. Lyon and T. Y. Ismael. Toronto: Macmillan of Canada, 1976, pp. 133-161.
- Sabourin, Louis. "Les programmes canadiens de coopération avec les États de l'Afrique (particulièrement avec l'Afrique francophone)." *Etudes Internationales*, 1 (décembre 1970), pp. 73-87.
- Sabourin, Louis. "Québec Courts Francophone Africa." Africa Report, 15 (April 1970), pp. 16-7.
- "Unconditional Aid to the Congo." External Affairs, 12 (October 1960), pp. 802-6.

2.4 Latin America

- "Aid for Latin America." External Affairs, 19 (June 1967), pp. 208-11.
- Belisle, J. Denis. "The Inter-American Development Bank." Foreign Trade, 133 (January 31, 1970), pp. 26-9.
- Brecher, Irving; and Brecher, Richard A. "Canada and Latin America: The Case for Canadian Involvement." *Queen's Quarterly*, 74 (Autumn 1967), pp. 462-71.
- "Canada and Inter-American Development Bank." External Affairs, 17 (January 1965), pp. 27-8.
- Canada and Latin America: Partners in Development. Toronto: Canadian Association for Latin America, 1974, pp. 4-5.
- Canada, Department of External Affairs. Preliminary Report of the Ministerial Mission to Latin America: October 27-November 27, 1968. Ottawa: 1969.
- "Canadian Aid to El Salvador." External Affairs, 22 (June 1970), pp. 207-8.
- "Canadian Assistance to Peruvian Earthquake Victims." External Affairs, 22 (September 1970), pp. 313-4.
- Canadian International Development Agency. Canada's Development Assistance to Latin America. Ottawa: 1974 (Briefing Papers).
- Cermakian, Jean. "Géographie et circulation internationale des capitaux: le cas de l'aide extérieure du Canada à l'Amérique latine." La Revue de Géographie de Montréal, 23 (1969), pp. 67-74.

- "Chile Commemorates Canadian Disaster Aid." External Affairs, 13 (June 1961), pp. 224-5.
- de Bondt, John. "How Canada Helps South American Development." Canada Commerce, 137 (July 1973) pp. 31-2.
- Goodman, Eileen. "Mr. Hargreaves Goes to Brazil." Canadian Business, 44 (August 1971), pp. 18 ff.
- Grey, James John. "Canada and Latin America." World Today, 32 (October 1976), pp. 376-86.
- "Loan to Colombia for Pre-Investment Studies." External Affairs, 20 (June 1968), p. 273.
- "New Agreement on Financial Aid to Latin America." External Affairs, 17 (August 1965), pp. 361-2.
- Randall, Stephen J. "Canadian Policy and the Development of Latin America." A Foremost Nation: Canadian Foreign Policy and a Changing World. Edited by Norman Hillmer and Garth Stevenson. Toronto: McClelland and Stewart, 1977, pp. 202-29.
- "Reconstruire le Guatemala." Coopération Canada, 26 (1977), pp. 15-8 (Eng. "Guatemala Rebuilds").

Chapter 3

Official Publications

3.1 Canadian Government

In this section are listed the serial publications of the various departments and agencies that have input into Canada's development assistance programs. The annual reports of these and other departments should also be consulted to locate information about their activities.

It should be recognized that there are other government departments that have input into Canada's development assistance programs. However, they have not been included because they do not have specific publications on the topic.

Specific articles by or about Canadian government will be found in either the General or appropriate Regional sections.

Department of External Affairs

The Department publishes *International Perspectives*, a bimonthly journal containing articles and a bibliography of recent publications on all aspects of Canada's foreign relations. This journal superseded *External Affairs* in 1971.

Statements and Speeches²

- Barton, W. H. "Canada at the United Nations." Winnipeg, May 13, 1977, p.3. (77/8)
- Baudouin, Marc. "Decolonization." New York, November 26, 1975. (75/39)
- Bégin, Monique. "More Canadian Cash for UNICEF." Ottawa, November 6, 1975. (75 / 45)
- Bégin, Monique. "More Canadian Money for U.N. Development Program." Ottawa, November 5, 1975. (75 / 44).
- Bourget, Maurice. "The Economic Development of Under-Developed Areas." Paris, November 27, 1951. (51 / 50)
- Cavell, R. G. Nik. "Asia and the Free World." Toronto, September 24, 1953. (53/38)
- Cavell, R. G. Nik. "Asia and the International Situation." Ottawa, May 21, 1952. (52/19)
- Cavell, R. G. Nik. "South East Asia and the Colombo Plan." Toronto, May 24, 1957. (57/29)
- Cavell, R. G. Nik. "The Colombo Plan." Toronto, December 4, 1952. (52/52)

²A chronological list of Statements and Speeches is found in Donald M. Page; Bibliography of Works on Canadian Foreign Relations, 1945-1970 and 1971-1975.

- Cavell, R. G. Nik. "Technical Assistance and the Colombo Plan." Toronto, October 5, 1951. (51/38)
- Croll, David. "Economic Development of Under-Developed Countries." United Nations, January 8, 1957. (57/5)
- Diefenbaker, John G. "Canada and the Commonwealth." London, England, November 4, 1958. (58/44)
- Diefenbaker, John G. "Peoples Want Peace, Not Propaganda." United Nations, September 26, 1960, pp. 7-9. (60 / 32)
- Diefenbaker, John G. "Western Policy Re-examined." Depaw University, Indiana, June 5, 1960, pp. 5-6. (60 / 21)
- Fleming, Donald M. "Organization for Economic Co-operation and Development." House of Commons, December 16, 1960. (61/1)
- Fleming, Donald M. "The OECD Takes Stock: 1962." House of Commons, December 3, 1962. (62/16)
- Fournier, Alphonse. "Methods of Financing the Economic Development of Under-Developed Countries." ECOSOC, Geneva, July 11, 1950. (50/29)
- Gérin-Lajoie, Paul. "Canada's Foreign Aid Priorities." Ottawa, February 4, 1971. (71 / 7)
- Gillespie, A. "Milestones on the Road to a New Economic Order." Paris, May 30, 1977. (77 / 11)
- Goldschlag, Klaus. "Canada and Latin America." Ottawa, October 19, 1972. (72/22)
- Goyer, Jean-Pierre. "Canada and Latin-America A Period of Mutual Discovery." Washington, D.C., June 30, 1970. (70 / 10)
- Green, Howard C. "Aid to India." House of Commons, January 19, 1960. (60 / 9)
- Green, Howard C. "Indus Waters Settlement." House of Commons, February 29, 1960. (60 / 14)
- Irwin, H. E. W. "Surplus Food and Hungry People." United Nations Second Committee, October 24, 1960. (60 / 35)
- Jamieson, Donald C. "Canada and Brazil: Enormous Potential for Economic Cooperation." Brasilia, Brazil, January 12, 1977. (77 / 1)
- Jamieson, Donald C. "The Commonwealth and World Security." Ottawa, September 20, 1977, p. 3. (77/14)
- Macdonald, Donald S. "The Role of the United Nations in International Development: The Canadian Contribution." Ottawa, May 3, 1967. (67/11)
- MacEachen, Allan J. "A Challenge of Historic Proportions." Paris, May 30, 1977. (77 / 10)
- MacEachen, Allan J. "Assessing Canada's Aid to the Third World." Port Hawkesbury, N.S., May 4, 1975. (75 / 14)
- MacEachen, Allan J. "Canada Adopts a New Development Assistance Strategy."
 Ottawa, November 6, 1975. (75 / 35)

- MacEachen, Allan J. "Canada and Africa." Toronto, February 19, 1975. (75/2)
- MacEachen, Allan J. "Canada and Egypt Strengthen Their Relations." Cairo, January 12, 1976. (76/1)
- MacEachen, Allan J. "Canada and Indonesia: The Dialogue Has Begun Well." Jakarta, Indonesia, August 25, 1976. (76 / 25)
- MacEachen, Allan J. "Highlights of Canada's Development Cooperation Strategy." Ottawa, November 6, 1975. (75 / 42)
- MacEachen, Allan J. "International Co-operation: Moral Imperative and Political Necessity." Ottawa, May 20, 1975. (75 / 15)
- MacEachen, Allan J. "On the Wake of the World Food Conference: Food Production and Rural Development." London, England, March 4, 1975. (75/5)
- MacEachen, Allan J. "Sharing and Survival." Nairobi, Kenya, May 7, 1976. (76/14)
- MacEachen, Allan J. "Strategy for International Development: 1975-80." Ottawa, September 2, 1975. (75 / 27)
- MacEachen, Allan J. "The Challenge of Change." New York, September 3, 1975. (75/26)
- MacEachen, Allan J. "The Commonwealth and Canada: Association For the Future." Toronto, November 27, 1974. (74/17)
- MacEachen, Allan J. "The Elimination of Man's Oldest Scourge and Most Persistent Enemy Hunger." Rome, November 6, 1974. (74/13)
- MacEachen, Allan J. "The General Policies of the Canadian International Development Agency." Ottawa, April 10, 1975. (75/9)
- MacEachen, Allan J. "UNCTAD IV: Important Stage on Road to New Economic Order." Ottawa, June 10, 1976. (76 / 19)
- Martin, Paul. "Aid Policies and Programmes." Washington, D.C., July 20, 1966. (66/32)
- Martin, Paul. "Aid Programmes and the Business Community." Toronto, September 14, 1966. (66/38)
- Martin, Paul. "Canada and Africa." Toronto, February 19, 1966, pp. 3-5. (66 / 8)
- Martin, Paul. "Canada and the Commonwealth Caribbean." Ottawa, December 8, 1970. (70 / 22)
- Martin, Paul. "Canada Forges Another Link with Latin America." Quito, Ecuador, May 10, 1972. (72/9)
- Martin, Paul. "Canada and 'La Francophonie'." Montreal, March 11, 1967, pp. 5-6. (67/7)
- Martin, Paul. "Canada's Contribution to Economic Development in the Less-Developed Countries." Columbia University, New York, April 28, 1967. (67/14)

- Martin, Paul. "The Challenge of Underdevelopment." Cleveland, September 26, 1964. (64/23)
- Martin, Paul. "Conditions for Peace in Africa and the World." Kingston, Ontario, June 17, 1966. (66 / 28)
- Martin, Paul. "Contributions to International Peace and Development." Toronto, August 26, 1967, pp.1-2. (67/27)
- Martin, Paul. "Development Aid and World Citizenship." Advertising and Sales Club, Hamilton, Ontario, January 6, 1964. (64 / 1)
- Martin, Paul. "Economic and Technical Assistance." United Nations, October 14, 1955. (55 / 36)
- Martin, Paul. "Educational Aid to French-Language Nations." Montreal, September 3, 1965. (65/21)
- Martin, Paul. "Food Problems of the Less Developed Countries." Washington, D.C., July 21, 1966. (66/33)
- Martin, Paul. "The Government, the People of Canada and Foreign Aid." Windsor, November 6, 1967. (67 / 34)
- Martin, Paul. "New Dimensions in Canadian Foreign Policy." University of New Brunswick, May 16, 1967, pp. 5-6. (67 / 17)
- Martin, Paul. "Overcoming the World Food Problem." Toronto, September 15, 1967. (67/29)
- Martin, Paul. "Progress in Co-operation Between Canada and French-Speaking Africa." Montreal, September 2, 1966. (66 / 36)
- Martin, Paul. "Some Reflections on the Principles Underlying Canadian Foreign Policy." Waterloo University, May 22, 1967, pp. 3-4. (67/18)
- Martin, Paul. "The United Nations Conference on Trade and Development." Geneva, March 24, 1964. (64/3)
- McKeen, S. S. "Fourth Technical Assistance Conference." United Nations, November 12, 1953. (53/45)
- Pearson, Lester B. "Brotherhood Between Nations The Colombo Plan." Toronto, February 20, 1956. (56/13)
- Pearson, Lester B. "Canada and the Colombo Plan." House of Commons, February 21, 1951. (51/6)
- Pearson, Lester B. "Canada and Technical Assistance to Underdeveloped Countries." Ottawa, March 4, 1952. (52 / 10)
- Pearson, Lester B. "Canadian Youth Serves the Developing Countries." Ottawa, October 1, 1965. (65/25)
- Pearson, Lester B. "The Colombo Conference." House of Commons, February 22, 1950. (50/6)
- Pearson, Lester B. "Debate on External Affairs." House of Commons, April 21, 1955, pp.3-4. (55/14)

- Pearson, Lester B. "New Aspects of International Competition." Toronto, March 13, 1956. (56/6)
- Pearson, Lester B. "Some Aspects of International Affairs." New Delhi, November 1955. (55 / 43)
- Pearson, Lester B. "The World We Live In." Toronto, June 27, 1954. (54/34)
- Pépin, Jean-Luc. "Canada's Trade with Developing Countries." Windsor, November 6, 1970. (70/16)
- Reid, Escott. "India's Development Programme." Ottawa, May 28, 1955. (55/16)
- Reid, Escott. "Our New Atlantic Partnership." Hanover, Germany, May 6, 1961, pp. 4-6. (61 / 6)
- Richard, Theogène. "Economic and Financial Questions in the United Nations."
 United Nations, October 9, 1957. (57 / 37)
- Robinson, H. B. "Canada Adopts a Multidimensional Approach to International Development." Ottawa, November 25, 1975. (75 / 43)
- Rogers, Benjamin. "Canada's Views on the United Nations Technical Assistance Programme." United Nations, November 5, 1957. (57 / 42)
- Rogers, R. L. "Canada and the Countries of the Pacific Basin." Ottawa, April 4, 1977, pp. 4-5. (77/7)
- Sharp, Mitchell. "Canada in the Global Community." Toronto, June 3, 1969. (69/11)
- Sharp, Mitchell. "Canada and the United Nations, 1968." United Nations' General Assembly, October 9, 1968, pp. 6-7. (68 / 15)
- Sharp, Mitchell. "Canada in a New World." Vancouver, January 17, 1972. (72 / 3)
- Sharp, Mitchell. "Canada in the World Community." Toronto, January 14, 1972, pp. 6-8. (72/2)
- Sharp, Mitchell. "Canadian Foreign Policy and the Third World." Toronto, September 18, 1970. (70/12)
- Sharp, Mitchell. "Independence and Interdependence: Principal Objects of Canada's Foreign Policy." St. John's, Newfoundland, February 25, 1972. (72/6)
- Sharp, Mitchell. "The Major Aims of Canadian External Policy." Ottawa, March 19, 1974, pp. 12-15. (74/3)
- Sharp, Mitchell. "The Major Concerns of Canadian Foreign Policy." Ottawa, May 15, 1973, pp. 7-9. (73 / 15)
- Sharp, Mitchell. "The Problems of International Economic Relations." New York, April 11, 1974. (74/5)
- Sharp Mitchell. "Relief for Pakistani Refugees in India." Ottawa, November 17, 1971. (71/27)
- Sharp, Mitchell. "The Role of Middle Powers in a Changing World." Ottawa, February 20, 1969, pp.6-7. (69/3)

- Sharp, Mitchell. "Some Aspects of the World Where Canada Works and Trades." Vancouver, May 11, 1972, pp. 2-3. (72/11)
- Sharp, Mitchell. "Visit to Africa." Ottawa, March 22, 1971. (71/12)
- Smith, Arnold. "Reflections on Neighbourhood." Toronto, November 22, 1954. (54/55)
- Smith, Sidney E. "The Colombo Plan: Programmes and Problems." Seattle, November 11, 1958. (58/45)
- Smith, Sidney E. "Facets and Form of Canada's Foreign Policy." Ottawa, February 12, 1958. (58 / 7)
- St. Laurent, Louis S. "Canada's Relations with Asia." Victoria, September 5, 1952. (52/33)
- Trudeau, Pierre Elliott. "The Character of Canada's Involvement With Latin America." Caracas, Venezuela, January 30, 1976. (76/11)
- Trudeau, Pierre Elliott. "Mexico-Canada Association Harmonious and Effective." Mexico City, January 23, 1976. (76/9)
- Winters, Robert H. "International Development: A New Dimension for Canadian Business." York University, November 6, 1967. (67 / 36)

Department of Industry, Trade and Commerce

The Department issues a quarterly journal, *Canada Commerce*, which has been indexed for all relevant articles. This journal superseded *Foreign Trade* in 1971.

External Aid Office

Superseded by the Canadian International Development Agency in 1967.

Report on Canada's External Aid Programs. Ottawa: 1962-1967. Annual reviews issued in 1961, 1962, 1963, 1964, 1965-66, and 1967. Superseded by the Annual Review of the Canadian International Development Agency beginning in 1967-68.

Parliament

Canada, Parliament, House of Commons. House of Commons Debates: Official Report.

The House of Commons and Senate Debates have not been exhaustively indexed as each sessional volume contains an index, and topics in international development are listed under such headings as:

Canadian International Development Agency External Affairs

With the exception of the special reports listed below, the *Minutes of Proceedings and Evidence* of the House of Commons Standing Committee on External Affairs and National Defence and the Senate Standing Committee on Foreign Affairs have not been exhaustively indexed as there are detailed indexes at the close of each session.

- Canada, Parliament, House of Commons, Standing Committee on External Affairs and National Defence. Report of the Subcommittee on International Development Assistance. 28th Parliament, 3rd Session, 1970-71, No. 29. Ottawa: May 27, 1971. Chairman: Mr. Georges Lachance.
- Canada, Parliament, House of Commons, Standing Committee on External Affairs and National Defence. *Minutes of Proceedings and Evidence*. 30th Parliament, 1st Session, 1974-75-76, No. 34. First Report of the Subcommittee on International Development. Ottawa: April 1976.
- Canada, Parliament, House of Commons, Standing Committee on External Affairs and National Defence. *Minutes of Proceedings and Evidence*, 30th Parliament, 2nd Session, 1976-77, No. 15. First Report of the Subcommittee on International Development. "The Debt Problems of Developing Countries." Ottawa: 1977.
- Canada, Parliament, House of Commons, Standing Committee on External Affairs and National Defence, Subcommittee on International Development. *Minutes of Proceedings and Evidence*. 30th Parliament, 1st Session, Nos. 1-22, July 22, 1975—June 17, 1976; 2nd Session, Nos. 1-11, March 30, 1977—June 9, 1977. Ottawa: 1975-77. Chairman: Mr Maurice Dupras.

Canadian International Development Agency

- Cooperation Canada magazine on international development (monthly, bilingual).
- Contact monthly newsletter on CIDA projects and other international development activities (bilingual).
- Action quarterly bilingual tabloid on the work of voluntary groups (non-governmental organizations) for world development.
- Development Directions; a Newsmagazine on International Development. (monthly). Superseded Cooperation Canada and Contact in May / June 1978.
 - Also available in French: Directions du Développement.
- Canadian International Development Agency. Annual Aid Review. Reports on Canada's development assistance program presented to the Development Assistance Committee of the Organization for Economic Cooperation and Development. Ottawa: 1971-.
 - Also available in French: Revue annuelle de l'aide.
- Canadian International Development Agency. Annual Review, 1967/68-. Ottawa: Queen's Printer (Information Canada), 1968-.
 - Continues External Aid Office, A Report on Canada's External Aid Programs.
 - Also available in French: Revue annuelle.
- Canadian International Development Agency. Non-Governmental Organization Division. Contributions to projects undertaken through Canadian Non-Governmental Organizations. Ottawa: 1972.

Export Development Corporation

News Release / Communiqué (irregular) EDC News (bimonthly)

Export Development Corporation. Annual Report, 1969-. Ottawa, 1970-.

International Development Research Centre

The following periodicals are available from IDRC. They focus on activities of the Centre and scientific research being carried out in developing countries.

IDRC Reports (quarterly)

(Also available in French: Le CRDI Explore, and Spanish: CIID Informa)

IDRC Annual Report 1970-71- . Ottawa: 1971- . (IDRC-003 / 71 ... e,f) Bilingual

IDRC Review 1974-75-. Ottawa: 1975-. (IDRC-054e, f, s, -.)

IDRC - The First 200 Projects.

Ottawa: 1975. (IDRC-032e).

Also available in French (IDRC-032f) and Spanish (IDRC-032s).

Supplements:

Projects 1975. Ottawa: 1975. (IDRC-047e).

Also available in French (IDRC-047f) and Spanish (IDRC-047s).

Projects 1976. Ottawa: 1976. (IDRC-072e).

Also available in French (IDRC-072f) and Spanish (IDRC-072s).

Projects 1977. Ottawa: 1977 (IDRC-097e).

Also available in French (IDRC-097f) and Spanish (IDRC-097s).

3.2 International Organizations

These are international organizations to which Canada belongs. None of the publications in this section deal exclusively with Canada's contribution to development assistance funds, but rather include Canada along with other contributors. The organizations listed here are the major ones involved in initiating development projects and disbursing multilateral contributions.

Specific articles by or about international organizations will be found in either the General or appropriate Regional sections.

Colombo Plan Bureau

Annual Report, 1952-. Newsletter (monthly)

Colombo Plan for Cooperative Development in South and Southeastern Asia.

Annual Report of the Consultative Committee, 1952-. Colombo: 1953-.

Each volume includes chapters on the contributors, including Canada.

The Commonwealth

Canada's major contribution to multilateral assistance programs is made through the Commonwealth Fund for Technical Cooperation (CFTC). Commonwealth trends, events, and activities are recorded in:

Commonwealth Currents (bimonthly)

Other reports of interest include:

Flow of Intra-Commonwealth Aid, 1968-. London: 1970-.

Annual Publication, chapter III includes countries supplying assistance, country policies, and multilateral assistance.

Report of the Commonwealth Secretary-General, 1965-. London: 1966-.

La Francophonie

Most multilateral support for Francophone programs has been channelled through the Fond international de coopération universitaire (FICU).

More modest funds have been given to the Agence de coopération culturelle et technique, which publishes the following periodical:

Agecoop Liaison (monthly)

International Monetary Fund

Finance and Development (quarterly) IMF Survey (24 issues per year)

International Monetary Fund. Summary of Proceedings of the ... Annual Meeting ... Washington, D.C.: 1946 - .

Other publications are listed in:

Catalogue of Publications, 1946-71 and subsequent annual lists.

Organisation for Economic Cooperation and Development Development Assistance Committee

 \dots Set up to secure an expansion of the volume of financial resources made available to the developing countries and to improve their quality \dots 3

Development Cooperation. Efforts and Policies of the Members of the Development Assistance Committee. Paris.

Each volume indicates Canada's official development assistance, private financial flows and grants by private voluntary agencies (NGO's), and comparative figures for other DAC members.

For other publications see the OECD Catalogue of Publications.

Regional and Subregional Development Banks

African Development Bank

Canada contributes to the African Development Fund, which assists the Bank in contributing to economic and social development in member countries.

³OECD Catalogue of Publications, 1976, p. 23.

Andean Development Corporation

Canada subscribes to this development financing organ of the Andean Group.

Asian Development Bank

ADB News (quarterly) Annual Report, 1967-.

Caribbean Development Bank

Annual Report, 1970- .

Inter-American Development Bank

Annual Report, 1960-.

Economic and Social Progress in Latin America (annual)

IDB news, 1974-. (monthly)

Proceedings of the ... meeting of the Board of Governors, 1960-.

United Nations

In addition to the major development-oriented UN specialized agencies such as FAO, ILO, UNESCO, WHO, and WIPO, Canada distributes multilateral aid through many UN social and economic programs including the UNDP, UNHCR, UNICEF, UNFPA, UNRWA, UNITAR, and the joint UN/FAO World Food Programme. The UNDP is the largest recipient of Canadian multilateral assistance within the UN family.

A summary report of the activities of the UN can be found in the annual Yearbook of the United Nations, which also refers the reader to more detailed sources where appropriate. Many UN publications refer to Canada's development assistance; these can be accessed through:

United Nations Document Index, Vol. 1-24, January 1950 — December 1973. New York. (ST/LIB/SER.E/-).

Superseded by:

UNDEX; United Nations Document Index. Vol. 1 - , January 1970 - . New York. (ST/LIB/SER.I-)

Series A: Subject Index Series B: Country Index

Series C: List of Documents Issued

A useful guide to the structure, activities, and publications of the UN and affiliated intergovernmental organizations is:

Winton, Harry, N. M. Publications of the United Nations System; a Reference Guide. New York: Bowker, 1972.

In Canada, UN publications are available through:

⁴Details of Canada's financial contributions to the regular programs of the UN, its special funds, and nonfinancial agencies are to be found in *Canada and the United Nations* 1945-1975 (Department of External Affairs, 1977, Appendix I, p. 193).

United Nations Association in Canada 63 Sparks St.
Ottawa, Ont. K1P 5A6

and

Renouf Publishing Co. Ltd. 2182 St. Catherine Street, W. Montreal, Quebec, H2H 1M7

United Nations Specialized Agencies

Food and Agriculture Organization of the United Nations

Ceres: FAO Review (bimonthly)

Report of the ... Session of the Conference, 1945 - . Rome: 1946 - .

FAO publications are indexed in FAO Documentation: Current Bibliography, 1967 - . (Title varies slightly)

International Labour Office — ILO

International Labour Review (twice monthly)

Official Bulletin (3 per annum)

ILO Information (bimonthly)

Publications available in Canada through:

ILO - Canada 178 Queen Street Ottawa, Ont.

United Nations Educational, Scientific and Cultural Organization - UNESCO

UNESCO Courier (monthly)

UNESCO Chronicle (monthly)

World Health Organization — WHO

World Health (monthly)

WHO Chronicle (monthly)

World Intellectual Property Organisation — WIPO

... to improve access to inventions, technology and trademarks for the developing countries.⁵

Industrial Property

World Bank Group. International Bank for Reconstruction and Development and its Affiliates

The largest recipient of Canadian multilateral aid is the International Development Association (IDA), which is the arm of the World Bank that lends ⁵Europa Yearbook, 1976, V.I. p.61.

money on concessional terms to the least developed countries.

Finance and Development (quarterly)

- International Bank for Reconstruction and Development. Annual Report, 1946 . Washington, D.C.
- International Bank for Reconstruction and Development. Statement of Loans. Washington, D.C.: 1947 .
- International Development Association. Statement of Developing Credits. Washington, D.C.: 1961 . (quarterly)
- International Finance Corporation, Summary of Proceedings, Washington, D.C.

Free and sales publications are listed in the World Bank Catalogue of Publications. (annual)

Chapter 4

Bibliographies and Reference Works

- Blanchette, Arthur E., (ed.). Canadian Foreign Policy 1955-65: Selected Speeches and Documents. Toronto: McClelland and Stewart in association with the Institute of Canadian Studies, Carleton University, 1977. (Carleton Library, No.103)
- Canada Year Book: An Annual Review of Economic, Social and Political Development in Canada. Ottawa: Department of Supply and Services, 1906 -
 - Each volume contains a report of Canadian international activities including aid contributions.
- Canadian Annual Review. 1960-70. Toronto: University of Toronto Press, 1961-71. Superseded by the Canadian Annual Review of Politics and Public Affairs.
- Canadian Annual Review of Politics and Public Affairs. 1971 . Toronto: University of Toronto Press, 1972 .
 - An annual survey since 1971 with each volume containing sections by R. B. Byers on the Third World and international development assistance. Supersedes the *Canadian Annual Review*, 1960-1970.
- Canadian Council for International Cooperation. Directory of Canadian Non-Governmental Organisations Engaged in International Development Assistance. Ottawa: 1974.
 - Specific publications by or about the various NGO's are listed in the appropriate sections of this Bibliography.
- Canadian Council for International Cooperation. Listing of International Development Projects and Activities Supported by Canadian Non-Governmental Organisations. Ottawa: 1972.
- Devindex Canada: Index to ... Canadian literature on economic and social development. Index de la litterature canadienne de ... sur le développement économique et social. Ottawa: International Development Research Centre, 1976 (IDRC-079, 098, -e,f).
- Granatstein, J. L.; and Stevens, Paul. Canada Since 1867: A Bibliographical Guide. Toronto: Hakkert, 1974, pp. 33-55.
- International Canada. Toronto: Canadian Institute of International Affairs and the Parliamentary Centre for Foreign Affairs and Foreign Trade, 1970 .
 - A monthly compilation of current events in Canada's international relations based on surveys of the press and government documents and including many brief articles on and references to international development assistance. Check the annual indexes under "International Development Assistance" and "Canadian International Development Agency." *Inter-*

- national Canada supersedes The Monthly Report on Canadian External Relations, 1962-1969.
- MacKay, Robert Alexander, (ed.). Canadian Foreign Policy 1945-1954: Selected Speeches and Documents. Toronto: McClelland and Stewart, 1970, pp. 364-72.
- Motiuk, Laurence; and Grant, Madeline. A Reading Guide to Canada in World Affairs. Toronto: Canadian Institute of International Affairs, 1972.
- Organisation for Economic Co-operation and Development. *Development Aid:*OECD-ICVA Directory of Non-Governmental Non-Profit Organisations.
 Paris: Organisation for Economic Co-operation and Development, 1967, pp. 102-63. Project Co-ordinator: Edwin Eggins.
- Overseas Institute of Canada. Directory of Canadian Agencies and Organisations in Overseas Aid. Ottawa: May 1963.
- Overseas Institute of Canada. *Directory of Canadians With Service Overseas*. 2nd ed. A listing of Canadians having experience in technical and educational assistance to the developing countries. Ottawa: 1966.
- Page, Donald M. A Bibliography of Works on Canadian Foreign Relations 1945-1970. Toronto: Canadian Institute of International Affairs, 1973, pp. 173-84.
- Page, Donald M. A Bibliography of Works on Canadian Foreign Relations 1971-1975. Toronto: Canadian Institute of International Affairs, 1977, pp. 94-8.

Author and Institutional Index*

Abbas, S. A. 33

45-50

710003, 5. 71. 55	15 50
Adams, David G. 33	Canada, Dept. of Industry, Trade & Commerce, Inter-
African Development Bank 53	national Financing Branch 12, 50
Andean Development Corporation 54	Canada, External Aid Office 12, 12, 16, 33, 50
Anderson, Bob 33	Canada, Parliament, House of Commons 50
Anglin, Douglas G. 9, 40	Canada, Parliament, House of Commons Standing
Angus, H. F. 33	Committee on External Affairs and National De-
Armstrong, Gregory 9	fence 22, 23, 28 , 51
Arsenau, Berchmans 42	Canada, Parliament, Senate Standing Committee on
Asher, Robert E. 24	Foreign Affairs 34, 39
Asian Development Bank 32, 54	Canadian Broadcasting Corporation 15
Asian Development Bank 32, 34	Canadian Church Observers 13
Baker, Gordon A. 41	Canadian Council for International Cooperation 13, 57
Bame, Kwabena N. 41	Canadian Executive Service Overseas 13, 14, 18, 23,
Barber, Clarence L. 9	28
Barkway, Michael 9	Canadian Institute of International Affairs 9, 13, 16,
Barrett, F. Michael 9	18, 26, 57, 58
	Canadian Institute on Public Affairs 15, 27, 30, 36
Barton, W. H. 45	Canadian International Development Agency 9, 13-
Baudouin, Marc 10, 45	14, 13, 14, 15, 16, 17, 17, 18, 18, 19, 19, 22, 22, 23,
Bauer, P. T. 10	23 , 24, 27, 27 , 28 , 30, 33 , 34, 35 , 37, 39, 41, 42, 43,
Beamish, Royd E. 10	47, 51
Becker, Jane 10	Canadian Labour Congress 14, 15
Bégin, Monique 45	Canadian National Commission for UNESCO 37
Belisle, J. Denis 43	Canadian News Synthesis Project 14
Belshaw, Cyril S. 10, 33	Canadian University Service Overseas 14, 18, 23, 48
Bentley, C. F. 33	Caribbean Development Bank 39, 54
Berke, Robert 42	Carleton University 21
Berrington, P. D. 33	Carr-Gregg, John Ralph Edward 35
Berry, Glyn R. 39	Cavell, R. G. Nik 14, 35, 45-46
Best, Jack 10	Cermakian, Jean 14, 43
Blanchette, Arthur E. 57	Chaplin, C. E. 35, 40
Bland, C. L. 42	Charles, K. J. 15
Bled, Cynthia 10	Chaussé, Denise 40
Bonin, Bernard 33	Chaussé, Jean-Guy 40
Bourget, Maurice 45	Clarkson, Stephen 15, 35
Brault, Marcel 10	Cohen, Maxwell 35
Brecher, Irving 10, 33, 43	Cohn, Theodore 15
Brecher, Richard A. 43	
Brewin, F. Andrew 10	Colaço, Francis X. 15
Briant, Peter C. 10	Colombo Plan 9, 33, 34, 35, 36, 37, 38, 45, 46, 47, 48,
British Columbia, Ministry of Agriculture 10	50
Bryce, Murray D. 33	Colombo Plan Bureau 35, 52
Burn, Henry Pelham 10	Commonwealth Secretariat 15, 21, 46, 47, 52-53
Bursey, M. B. 11	Council on Foreign Relations 27
Burton, Ian 11	Cox, Terrance 15
	Croll, David 46
Byers, R. B. 11, 57 Byers, T. J. 11	Culhane, Claire 36
Dycio, 1. J. 11	Cunningham, George 15

Canada, Dept. of Agriculture 40 Dacre, Douglas 36 Canada, Dept. of External Affairs 12, 33, 34, 42, 43, de Bondt, John 15, 16, 44

^{*}Note: Items are referred to by page number. Those page numbers in boldface refer to articles *about* the person or institution.

Devlin, Edward 13, 16 DeWolf, Kenneth G. 36 Diefenbaker, John G. 24, 36, 46 Dobell, Peter C. 16 Dobell, W. M. 16 Doupe, Jack H. 16 Dudley, Leonard 16 Duchesne, Paul G. 14 Dupuis, Michel 36

Eayrs, James 16 Edmondson, Locksley G. E. 39 Elman, Russell 36 Export Development Corporation 52 Eyton, A. T. 36

Fatouros, Arghyrios A. 16 Fairweather, Gordon 16 Finkelstein, Rytsa 39 Fischer, Lewis A. 16 Fitz-Earle, Malcolm 9 Fleming, Donald M. 46 Fletcher, Mary 16 Fontaine, F. 17 Food and Agriculture Organisation of the United Nations 12, 55 Fournier, Alphonse 46 La Francophonie 42, 47, 48, 53 Franck, Thomas 41 Fraser, Colin 17 Fraser, R. Blair 17, 36 Fraser, Duncan G. L. 39 Fréchette, Pierre 19 Fulford, Robert 17

Garry, Robert 17 Gérin-Lajoie, Paul 17-18, 24, 36 46 Gillespie, A. 46 Gillis, Frank 18 Glyde, Henry 18 Godfrey, Dave 23 Goldfarb Consultants Ltd. 18 Goldschlag, Klaus 46 Goodman, Eileen 44 Gordon, J. King 18 Gordon, Sheldon 18 Gover, Jean-Pierre 46 Grafftey, W. Heward 18 Granatstein, J. L. 57 Grant, Madeline 58 Green, Howard C. 46 Gregory, John B. 18 Grenier, Raymond 42 Grey, James John 44 Grey, Rodney Y. 18

Hamlin, D. L. B. 27 Harker, John 19 Harrison, W. E. B. 19 Helleiner, Gerald K. 19 Henry, Jacques 42 Heyneman, Donald 19 Higgins, Benjamin 19 Higgins, Jean 19 Hillborn, Kenneth 19 Hillmer, Norman 40, 44 Hitschmanova, Lotta 19 Holmes, John W. 19, 27 Hopper, W. David 19-20 Huff, W. G. 20 Hulse, Joseph H. 20 Humphreys, David 20 Hymer, Stephen 20

Innis, Hugh 20
Inter-American Development Bank 43, 54
International Bank for Reconstruction and Development 56
International Development Association 56
International Development Research Centre 10, 14, 19, 20, 20, 26, 27, 28, 29, 30, 31, 52, 57
International Finance Corporation 56
International Labour Office 55
International Monetary Fund 53
Irwin, H. E. W. 46
Irwin, John 37
Islam, Nasir 20
Ismael, Tareq Y. 22, 28, 38, 39, 41, 43

Jamieson, Donald C. 46 Jeanes, C. W. L. 21 Jecchinis, C. 28 Johnson, Harry G. 21 Juvet, C. S. 21

Kaplansky, Kalmen 21 Kavic, Lorne 21, 37 Keenleyside, Hugh L. 21 Keirstead, B. S. 21 Kelson, Robert N. 16 Keyfitz, Nathan 21, 37 Kidd, J. R. 21 Kingsmill, C. G. 37

Latin American Working Group 21
Laird, Marshall 21
Langdon, Steven 40
Leach, Richard H. 21
Leavitt, Donald H. 39
Lebel, Ronald 21
Legum, Colin 40
Lemelin, Claude 42
Leroy, Marcel 21
Levitt, Kari 39
Linton, Neville 22, 41
Lloyd, Trevor 22
Lotz, Jim 22, 40
Lower, J. Arthur 376

Lyon, Peyton V. 22, 28, 38, 39, 41, 43 Ma, Nancy 37 McConkey, David 22 Macdonald, Donald S. 22, 46 McDonald, Mary L. 14 Macdonald, R. St. J. 28 MacEachen, Allan J. 22, 46-49 McGill University, Centre for Developing Area Studies 39 McInnis, Edgar 23 McIntyre, Alister 39 MacKay, R. A. 28, 37, 58 McKeen, S. S. 48 MacKenzie, A. E. D. 23 McKinnell, Robert 23 McLean, Ross 37 MacLean, Logan 23 McLoughlin, Peter F. M. 23 MacQuarrie, Heath 39 McWhinney, Bill 23 Magee, David 23 Maione, Romeo 23 Malhotra, S. P. 23 Marion, Jean Claude 31 Marsh, Reid 37 Martin, Paul 23-24, 37, 40, 42, 47-48 Martin, Robert 24 Mason, Edward S. 24 Masters, Donald C. 24 Matthews, Robert 40 Matthews, Robert J. 41 Matthews, Robert O. 40 Matthews, Roy A. 24 Mboya, Tom J. 24 Meagher, M. 24 Mehling, Jean 37 Mende, Tibor 24 Meredith, Brian 24, 43 Mezei, Kathy 33 Midy, Franklin 39 Mills, Joseph C. 24 Mishra, Sudhakanta 38 Montmarguette, Claude 16 Moran, H. O. 24, 38, 39, 41, 43 Morisette, Brigitte 24 Morris, Robert C. 24 Morrison, Barrie M. 38 Motuik, Laurence 58 Nash, C. Knowlton 24 Social Development 25

Lucas, R. D. 22

Lux, André 22

National Committee on Canada's Participation in North-South Institute 25

Ohlin, Goran 25 Organisation for Economic Co-operation and Development 15, 25, 46, 53, 58

Overseas Development Institute 15 Overseas Institute of Canada 25, 58

Page, Donald M. 58 Pal, D.-D. 25 Parlour, Roger R. 38 Pearson, Lester B. 26, 26, 48-49 Pelletier, Irénée 26 Pépin, Jean-Luc 49 Peters, Stuart 26 Piché, Raymond 26 Plumptre, A. F. W. 26 Pogson, T. W. S. 26 Porrier, Rolland P. 40 Preston, Richard A. 26 Private Planning Association 24, 39 Proudfoot, Stuart B. 24

Randall, Stephen J. 26, 44 Reid, Escott 26, 38, 49 Reuber, Grant L. 27 Richard, Theogène 49 Richardson, Ken 27, 38 Ritchie, A. E. 28, 37 Robb, William 27 Robert, Marika 40 Robinson, H. B. 27, 49 Roblin, Duff 48 Roche, Douglas 48 Rogers, Benjamin 49 Rogers, R. L. 49 Rothwell, B. E. 38 Royal Institute of International Affairs 16 Rubin, Seymour J. 27 Ruddick, Valley 27

Sabourin, Louis 27-28, 43 St. Laurent, Louis S. 50 Sallery, Robert D. H. 14 Sanger, Clyde 28, 40 Sarbadhikari, P. 28 Schiele, Robin 28 Scott, F. R. 28 Sharp, Mitchell 28, 49-50 Shute, J. C. M. 40 Sir George Williams University 28 Smith, Arnold 50 Smith, Evan 38 Smith, Sidney E. 50 Soward, F. H. 29 Spencer, Robert A. 29 Spicer, Keith 29, 38 Spurgeon, David 29 Stabler, Ernest 29 Steen, Jim 41 Stein, David Lewis 29 Stevens, Paul 57 Stevenson, Garth 40, 44 Strong, Maurice 29 Sutherland-Brown, Malcolm 29 Swanson, Roger F. 30

Taylor, Rupert J. 29 Tepper, Elliot 38 Thapa, B. B. 30 Thomson, Dale C. 30, 38 Triantis, Stephen G. 30 Trudeau, Pierre Elliott 30, 50

United Nations Association in Canada 30, 55
United Nations 12, 21, 22, 28, 30, 45, 46, 49, 54
United Nations Children's Fund (UNICEF) 12, 45
United Nations Conference on Trade and Development
(UNCTAD) 16, 24, 47, 48
United Nations Development Programme (UNDP) 22, 30, 45
United Nations Educational Scientific and Cultural
Organization (UNESCO) 37, 55
United Nations Relief and Works Agency for Palestine
Refugees in the Near East (UNRWA) 30
University of Ottawa, Institute for International Cooperation 20, 29, 41
Usher, Dan 30

Van Arkadie, Brian 20 van Praagh, David 38 Virtue, David 31

Wagenberg, R. H. 29, 31
Walmsley, Norma 31
Warren, J. H. 38
Westall, Stanley 31
White, Charles A. 31
Williams, Bruce M. 31
Williams, Peter 31
Winters, Robert H. 50
Winton, Harry N. M. 54
Woodcock, George 39
World Bank Group 24, 55-56
World Health Organization 12, 55
World Intellectual Property Organisation 55
Wright, Hume 31

Zagorin, Ruth K. 31

CARLETON UNIVERSITY

THE NORMAN PATERSON SCHOOL OF INTERNATIONAL AFFAIRS

Titles in the Bibliography Series

(Items marked * are out of print and xeroxed copies may be obtained by writing directly to the Inter-Office Loans Department of Carleton's Main Library.) \$0.20 a sheet.

- *1. Abonyi, Arpad; and Campbell, Bruce. International Integration; A Select Bibliography, December 1975, \$1.50.
- 2. Ervin, Linda. Council for Mutual Economic Assistance; A Selective Bibliography, December 1975, \$1.50. (prepaid)
- 3. Kelly, Michael J.; and Mitchell, Thomas H. Violence, Internal War and Revolution; A Select Bibliography, April 1976, \$2.00. (prepaid)
- 4. Thomson, Jennifer. The Law of the Sea With Special Reference to Canada; A Select Bibliography, September 1976, \$1.50. (prepaid)
- 5. Mandlowitz, Jason E.; Abonyi, Arpad; and Shastri, Pradip. Approaches to the Study of International Relations; A Select Bibliography. September 1977, \$2.00. (prepaid)

